THE POLITICAL ECONOMY OF DRONES

by

Abigail R. Hall

Dissertation Submitted to the Graduate Faculty of George Mason University in Partial Fulfillment of The Requirements for the Degree of Doctor of Philosophy Economics Committee: Director Department Chairperson Program Director Dean, College of Humanities and Social Sciences Date: _____ Spring Semester 2015 George Mason University Fairfax, VA

The Political Economy of Drones

A Dissertation submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy at George Mason University

by

Abigail R. Hall Master of Arts George Mason University, 2013

Director: Christopher J. Coyne, Professor Department of Economics

Spring Semester 2015 George Mason University Fairfax, VA



This work is licensed under a creative commons attribution-noderivs 3.0 unported license.

DEDICATION

For my sister Betsy, I miss you every day.

ACKNOWLEDGEMENTS

I thank those who have worked so selflessly on my behalf, provided me with opportunities to learn, and assisted my growth as a teacher and scholar. I thank my advisor, mentor, and dear friend, Christopher J. Coyne for his wisdom, encouragement, and saintly patience. I thank Peter J. Boettke and Peter T. Leeson for their lessons, counsel, encouragement, and feedback on this any countless other projects. I am truly indebted. I thank Virgil Storr, the Mercatus Center at George Mason University and the F.A. Hayek Program in Philosophy, Politics, and Economics for continuous support and opportunities. I thank Tyler Cowen for his feedback, encouragement, and chances to explore new topics.

I am grateful to the Institute for Humane Studies for their support on a variety of margins. I thank the Independent Institute for the opportunity to work with them and for providing me an outlet for my work.

I thank my Mom and first teacher, Mary Hall. I thank my Dad, Chet, for acting as my first (and still wonderfully demanding) editor and critic. Thank you for the years of love and sacrifice. Thank for instilling a love of mankind and the love of learning. Thank you for all the trips to the library, watching of History Channel documentaries, and encouraging my passions. I thank my brother, Joe Hall, for *never* taking me at my word and forcing me to produce evidence. I thank Edgar A. Blanco for his love and encouragement.

I wish to thank my two undergraduate mentors, Myra McCrickard and Frank Raymond. Without their guidance and teaching, I may have never discovered my passion for economics.

Finally, but importantly, I thank my friends and colleagues Joy Buchanan, Eric Cellar, Erica Cellar, Thomas K. Duncan, Stefanie Haeffele-Balch, Jayme Lemke, Peter Lipsey, Kyle W. O'Donnel, Liya Palagashvilli, Jacob W. Russ, Alexander W. Salter, Solomon Stein, Paola Suarez, and Jessi Troyan. Each has provided me with support, feedback, and encouragement. Thank you for all you have done.

TABLE OF CONTENTS

List of Figures	Page
Abstract	viii
The Political Economy of Drones	1
1 Introduction	1
2 The Origins of Entanglement ⁻ 1900-1948	1
3 Modern Drones and the Maturation of Entanglement: Cold War—pre-9/11	12
4 The Post-9/11 Expansion of Drones and Entanglement	16
5 Conclusion	26
Drones: Public Interest, Public Choice, and the Expansion of Unmanned Aerial Veh	nicles 28
1 Introduction	28
2 Public Interest and Implications for Drone Policy	32
Conjecture 1: Defense expenditures on the production of UAV technology are allocated to maximize defense and security for U.S. citizens	33
Conjecture 1.1: Producing drones is the most cost effective means, relative to k alternatives, of achieving U.S. security objectives.	nown
Conjecture 2: Ineffective or counterproductive drone policies would be elimina modified	ted or 34
3 Data and Evaluation of the Public Interest Conjectures	37
Conjecture 1: Defense expenditures on the production of UAV technology are allocated to maximize defense and security for U.S. citizens	38
Conjecture 2: Ineffective or counterproductive drone policies would be elimina modified.	ted or 41
4 Public Choice as an Alternative Framework	45
4.1 Private Industry	47
4.2 Congress	52
5 Conclusion	57

Drones Come Home: Foreign Intervention and the Use of Drones in the U.S	59
1 Introduction	59
2 The Boomerang Effect	64
3 Drones as a Tool of Foreign Social Control	71
4 Drones as an Illustration of the Boomerang Effect	74
4.1 Arthur Cebrowski, Force Transformation, and Domestic Drone Use	74
4.2 The Rise and Return of the Predator	81
4.3 Implications of the Boomerang Effect: Domestic Drones and the Threat to Privacy and Liberty	85
5 Conclusion	89
References	91

LIST OF FIGURES

Figure	Page
Figure 1: DOD Spending on UA: 1995–2013 (in millions)	17
Figure 2: Lobbying Expenditures by the Top Five Drone Manufacturers, in Millions,	
2000–2012	48
Figure 3: Drone/UAV Contracts (in millions of dollars) by Year ³	50

ABSTRACT

THE POLITICAL ECONOMY OF DRONES

Abigail R. Hall, Ph.D.

George Mason University, 2015

Dissertation Chair: Dr. Christopher J. Coyne

This dissertation consists of three essays that examine the interplay between private and public actors in the evolution, development, and implementation of this technology. My work provides a political economy analysis of the drone industry through public choice economics. This serves as a clear demonstration of how the incentives faced by various private and public actors and the interactions between these groups work to influence decisions regarding the evolution of national defense and policies surrounding military technologies.

THE POLITICAL ECONOMY OF DRONES

1 Introduction

World governments spent more than \$6.6 billion on Unmanned Aerial Vehicle and Unmanned Aerial System (UAV, UAS), or "drone," technology in 2012. This number is expected to increase to \$11.4 billion a year over the next decade for a worldwide UAV market worth more than \$89 billion (PR Newswire 2012). While the number of drones currently in operation is unknown, the International Institute for Strategic Studies has identified 56 different types of UAVs in use in eleven different countries (Guardian 2012). Present estimates place the number of drones in use by the U.S. government above 7,000, compared to fewer than fifty in 2000 (Tomiuc 2012). Although the U.S. presently dominates the world in terms of drone use, China, France, Germany, India, Israel, India, and Russia among others, are either known or suspected to have stocks of UAVs and still other nations have expressed interest in purchasing the technology (Defense Industry Daily 2013).

Drones are a core component of U.S. military operations and their use has controversial implications both domestically and abroad (see Miller 2012, Boyle 2013). The rise of UAV technology has effectively lowered the cost of military engagements and allowed the U.S. to undertake significant military action in Pakistan, Yemen, and Somalia without declaring war or deploying a significant number of troops. However, controversy has emerged regarding the accuracy of this technology as means of targeted

warfare, as well as their efficacy, and ethical ramifications (Strawser 2010, Singer 2009). Questions regarding the identity of drone targets and the unknown number of civilian causalities have sparked further debate. Estimates of civilian casualties range from 150 to thousands (Roggio and Mayer 2013, New America Foundation 2013). According to one estimate, for every one target killed by a U.S. drone strike, fifty innocent people are also killed (Kilcullen and Exum 2009).

Further, there has been a strong push by domestic manufacturers of UAV technology for the U.S. government to relax many of the rules currently prohibiting drone sales abroad and preventing the use of drones in domestic airspace. Numerous government organizations have also pushed to expand drone use. The U.S. Air Force, Army, Navy, and Special Forces all use drone technology and have worked with private contractors to research, develop and test new drones for use domestically and abroad. The Department of Homeland Security (DHS) and the U.S. Customs and Border Protection agency (CBP) have sought to expand the use of drones along the U.S.-Mexico border (Costantini 2012). State and local police have also taken an interest in the technology. By July 20, 2012, the FAA had authorized 106 state and local government entities to fly drones in U.S. airspace (Jeffrey 2012). The push for expanded use has raised concerns over safety, privacy, and government abuse, prompting immense scrutiny from both sides of the political aisle with many state lawmakers introducing legislation that attempts to restrict or ban the use of drone technology domestically (Goodale 2013, Crump and Stanley 2013).

This chapter provides the first political economy analysis of drones through the lens of public choice economics. Ideally, decisions regarding the use of drones (and national defense more generally) would be made to protect the U.S. and to promote the general interests of U.S. citizens. However, in contrast to this "public interest" view, the "public choice" model predicts that decisions made by politicians regarding UAV technology will be driven by the incentives created by political institutions rather than by some higher ideal (see Buchanan 2003). The public choice perspective emphasizes that the payoffs attached to various courses of action result from the nature of political rules, and those acting within that system will respond accordingly to these payoffs (Brennan and Buchanan 1985). This means that an array of political influences, and not some higher ideal of the "public interest," will drive how the use of drones evolves.

When studying issues of defense, particular attention must be paid to the role of actors in both the polity and in private industry. In developed countries, such as the U.S., the government maintains a monopoly on defense but relies heavily on private industry for defense-related production. These public-private linkages influence each other and, in doing so, influence the trajectory of defense policy and production. Within this context, the goal of our analysis is twofold. First, we identify the initial formation of these entanglements, or network relationships, as they relate to UAV technology. Second, we trace how these linkages increase, strengthen and expand during times of crises to generate benefits for (both public and private) Big Players—actors whose decisions and exercise of power (monetary, political, etc.) are able to influence and shape the course of events (see Koppl and Yeager 1996 and Butos and Koppl 1993). Higgs (1987) and Smith,

Wagner, and Yandle (2011) emphasize that moments of crisis (real or perceived) encourage new interactions and expand previously existing linkages between Big Players in the polity and the market as the demand for political action breaks down traditional separations.

Our analysis contributes to three strands of literature. First, we extend the literature in public choice and political economy as it relates to entanglements between public and private actors (Higgs 1987; Wagner 2009a,b, 2013; Smith, Wagner, and Yandle 2011). Acemoglu and Robinson (2013) have recently emphasized the importance of appreciating political economy issues when designing policies. Given the current, and foreseeable, controversies over drones, a public choice analysis of the topic is important for informing policy discussions. Second, and related, we contribute to the literature on the "military-industrial complex." Research in this area focuses on the connection between the military, political, and private sectors (see Melman 1970, 1971; Adams 1981; Pursell 1972; Hossein-Zadeh 2006; Duncan and Coyne 2013a,b). We extend this literature by analyzing the intricacy, breadth, and strength of the relationships between the Big Players in the UAV industry. Understanding the history of this technology, as well as the major players in the industry today, is of central importance to designing effective policies. Third, we contribute to the literature on the military history of the U.S. (Weigley 1960; Millett, Maslowski, and Feis 2012, Williams 2013) by tracing how the emergence and evolution of UAV technology by the U.S. military.

The subsequent sections analyze the drone industry in the U.S. from its origins to the present and examine three distinct time periods. We use the terms "drone" and

"UAV" broadly to refer to any aircraft without a human pilot controlled either by remote control or autonomously via computer. Section 2 discusses the origins of the linkages between private and public actors in the UAV industry from 1900-1948. Its purpose is to provide context of the initial emergence of drone technology and early entanglements in the drone industry. Section 3 examines the Cold War through the pre-9/11 period. It was during this time that earlier linkages matured and modern drone technology emerged. Section 4 analyzes the post-9/11 period with particular focus on how this crisis created an opening for the expansion of the drone industry. Established entanglements worked to drive this progression and, in the process, subsumed other private and public actors. Section 5 concludes.

2 The Origins of Entanglement: 1900-1948

Unmanned aerial vehicle (UAV) technology evolved along with the aviation industry in the U.S. Indeed, the first attempt at creating an UAV occurred soon after the invention of the airplane in 1903. As early as 1913, the U.S. Navy provided funds for the development of a radio-controlled aircraft (Hunsaker 1954). By 1915, inventors Elmer Sperry and Peter Cooper Hewitt were working to develop a pilotless aerial torpedo—the first drone (Newcome 2004: 16).

In 1915, the U.S. Navy created the Naval Consulting Board (NCB), an organization dedicated to the creation of new technologies, organizing the process of receiving inventions from the public, and increasing the Navy's technological capabilities (Scott 1920: 3-4). Consisting of approximately two dozen private citizens in various industries, including Sperry and Cooper, the NCB had the effect of entangling the military with the private sector (including the infant drone industry). Through their connection, Sperry and Cooper obtained an audience with the Secretary of the Navy and were ultimately allocated \$200,000 (more than \$4.2 million in 2012 dollars) to further develop their unmanned aerial torpedo (Pearson 1969: 71).

It is at this juncture one finds the foundations of the mutually beneficial relationship between the government and private military contractors and observes the emergence of future Big Players in the drone industry. William E. Boeing, founder of the Boeing Airplane Company, for example, worked throughout the period to design and build aircraft. In 1917, having heard of the Navy's desire for planes, Boeing had one of his planes deconstructed, sent to naval offices in Florida, and reassembled for military testing. The company quickly received an order for fifty planes (PBS 2013a). Glenn L. Martin, (founder of the Glenn L. Martin Company, now Lockheed Martin), organized his own firm in 1918 building specialty aircraft—including a design for a military bomber (National Aviation 2013a). Leroy Grumman (Grumman Aeronautical Engineering) Company, now Northrop Grumman), served as a pilot in WWI and through the military had obtained a degree in aeronautical engineering (Fetherston 1998). After completing his education, the Navy stationed Grumman at Loening Aeronautical Engineering Corporation to oversee the construction of Navy aircraft (Thruelsen 1976: 21). Seeing the potential profits from military contracts, Grumman resigned his Navy commission in 1920 to work for Loening. His military ties would remain important, however, as Grumman started his own company (Thruelsen 1976: 36). It was also during the interwar period that John Northrop (founder of Northrop Aircraft, later Northrop Grumman) and

Allan and Malcolm Loughead (Lockheed Aircraft Company, later Lockheed Martin) began to develop their products which would later be used by the military.

Although the military produced many of its own weapons during the period, the private defense industry took active steps to influence and shape the trajectory of military production. In 1919, the American Defense Preparedness Association (ADPA) was created by the industry with the goal of "increas[ing] weapons technology, improv[ing] defense management, and maintain[ing] a strong science-industry defense team [which would be] continually responsive to all needs of the development, production, logistics, and management phase of national preparedness" (National Defense Industrial Association 2013). The ADPA and other organizations would become immensely important in the development of the UAV industry as private industry sought to strengthen its ties with the military.

A renewed emphasis on UAV technology emerged in 1935. Following a visit to the British Royal Navy and observing the British advancements in creating target drones, Admiral William Standley returned to the States and ordered the development of similar technologies for the American fleets (Newcome 2004: 63). The Army Air Corps Act of 1926 (AACA), which had created a set of rules for negotiating contracts between the military and the private sector, allowed the Navy to utilize private contracts in order to engineer the drones. In the mid-1930s, a division of Northrop Aircraft was contracted to complete the task of creating the UAVs. Although other companies were simultaneously contracted, it was Northrop's Radioplane Co. that would ultimately obtain success and create the OQ-2A target drone. The Navy then contracted the firm to produce an

additional 1,000 of the drones for use in antiaircraft gunnery training (National Museum 2011).

At the start of WWII in 1939, both the Navy and Army used the enhanced contracting abilities granted by the ACCA to develop and use unmanned aerial technology. These abilities were expanded again after President Franklin D. Roosevelt declared a state of emergency on December 8, 1939. Following the declaration, Congress passed the Navy Reconstruction Act, allowing the Secretary of the Navy to "negotiate contracts for the acquisition, construction, repair, or alteration of complete naval vessels or aircraft, or any portion thereof...with or without competitive bidding" (quoted in Brown 2005: 7). The War Powers Acts issued in the early 1940s were particularly important from the public choice perspective. These Acts altered the rules under which the government could contract with private firms making it easier for the military to bypass much of the formal acquisition process (Brown 2005: 8). Instead of contracting with companies that offered the lowest price, the military could now contract more freely with those companies with whom it had already established a relationship. It followed that as the military looked to develop more drones for reconnaissance, anti-aircraft training, and attack drones, that those companies which had preexisting relationships with the military, like Northrop's Radioplane Co., were awarded contracts.

The passage of these acts, combined with the demands of WWII worked to expand the defense and aviation industries and further solidified the relationship between the U.S. military and private defense contractors. In 1943, for example, the U.S. Army's Air Tactical Service Command (ATSC) initiated a deal with Lockheed Aircraft Corporation to develop and produce a new string of jet fighters (Lockheed Martin 2013c). Boeing had used its own funds beginning in 1938 to develop a new bomber. When the Air Corp announced its formal request and design specifications for bombers to manufacturers at the start of the war, Boeing responded quickly, obtained the contract, and, by 1942, had received an order for 500 aircraft (Bowers 1989: 319). Northrop would be commissioned by the Army to build more than 700 "Black Widows," aircraft which could find and destroy enemy targets in the dark and inclement weather (Smithsonian 2013b).

Not only did these exchanges provide the U.S. military with a cadre of advanced war machines, but they also worked to expand the defense companies immensely. To give but one example of how these government contracts impacted private contractors, consider that by 1945, Grumman had seen such an increase in business that it added thousands of people to its payroll. At its peak during the war, the company employed more than 20,500 individuals and produced more than 650 aircraft monthly (Thruelsen 1976: 218). This expansion matters because as the private defense industry grew, so too did its influence on lawmakers in Washington, D.C.

The war effort also expanded the demand for drones as evidenced by the number of UAV-related contracts the military granted to private firms during the war. The Army contracted with Radioplane Co. again in 1939 to manufacture what would become the OQ target drone series. The Navy would also contract the company for the technology and more than 15,000 of the drones were used to train American anti-aircraft gunners throughout the war (Newcome 2004: 58). McDonnell Aircraft (later a part of Boeing) was contracted to build an unspecified number of "Katydid Drones" in an effort to compete with the German V-1 rockets (Smithsonian 2013a). Boeing was commissioned to convert "war-weary" B-17 "Flying Fortress" bombers into radio-controlled assault drones (Parsch 2003). Ryan Aeronautical (now part of Northrop Grumman), Lockheed Aircraft Corporation, Glenn L. Martin Company, and Vought (now part of Northrop Grumman) all contracted with the U.S. Army Air Force, U.S. Air Force, or Navy during the war to design and manufacture various types of drones (Parsch 2010).

For our analysis, World War II was significant in that it ushered in what has been called a "permanent war economy"—a situation characterized by constant funding for military equipment and supplies in order to develop and amass enhanced military capabilities during times of both war and peace (see Melman 1985 and Duncan and Coyne 2013a,b). Whereas the previous concept of war was one of responding to immediate and actual threats and aggression (e.g. an attack on American ships, etc.), the new military ideal was to maintain a state of constant preparation for future potential conflict. One consequence of this constant preparation was the establishment of a more permanent relationship between government and the private defense industry. Duncan and Coyne (2013a: 2) describe how such an economy "set in motion a process…whereby private actors respond[ed] to the opportunities presented by a state of permanent war and adjust[ed] their behaviors to take advantage of new profit opportunities....[T]he private economy [became] increasingly intertwined with the state."

The rise of the permanent war economy had important implications for the entanglement between the private defense industry and the government. While the

experiences of the World Wars had allowed for many defense firms to further specialize their products (e.g. Grumman's continued production of naval technology), the emergence of the permanent war economy worked to increase the importance of networking and social relationships between contractors and the military for the acquisition of contracts. Higgs (2007: 308) argues that following the end of WWII, military contracts "came to turn not on price, but on technical and scientific capabilities, size, experience, and established reputation as a military supplier—vaguer attributes that are easier to fudge for one's friends." This implies that those firms who had received contracts and built political relationships with the government before the war were more likely to be the recipients of contracts after the war. Indeed, following the conclusion of WWII, we see how the entanglements established during WWI, the interwar years, and WWII increased, strengthened, and generated further benefits for the Big Players in the market and the polity. For example, Lockheed's services were retained by the military for the continued production of military aircraft after WWII. The Glenn L. Martin Company began ventures in commercial aircraft, but received contracts to produce missiles and rockets for the U.S. military (Lockheed Martin 2013c). Grumman, in keeping with its past connection to the U.S. Navy, obtained long-term contracts with naval forces to design and manufacture new combat aircraft (Treadwell 1990: 90, 99, 120).

Efforts were made by actors both in the U.S. military and in private firms to maintain and strengthen the relationships which had been forged during the war. In 1948, the ADPA changed its name to the American Ordnance Association in order to reflect the changes brought about by the creation of the Department of Defense and expanded its

activities to include all branches of the military. The National Security Industrial Organization (NSIO) was formed from the Navy Industrial Association during the period in an effort to "establish and foster a close working relationship and effective two-way communication between government, primary defense, and the industry which supports it" (NDIA 2013). The Association looked to influence relationships between the military and weapons manufacturers in areas of research and development, procurement, and many other areas (*Ibid*). Brown (2005: 8) describes the push from both the market and the polity to continue the contracting which had taken place in wartime.

[T]hese relationships had...produced some of the most important weapons of the war....In the years that followed...Congress and the president, at the behest of the armed forces and representatives of the industry, drafted new laws to improve military procurement and encourage further cooperation between private industry and the government, with the understanding that such collaboration would strengthen the armed forces and yield even more impressive and powerful weapons.

At this point, the entanglements underpinning the current drone industry were

well established. These relationships would further mature during the Cold War period.

3 Modern Drones and the Maturation of Entanglement: Cold War—pre-9/11

The sustained threat of the Soviet Union further worked to expand the relationship between the military and private defense industry and increased interest in more advanced UAV technology. Although drones in the past were used almost solely for target practice and training purposes, the military also saw UAV technology as a potential tool for reconnaissance missions. Throughout the Cold War period, "radioplanes" were manufactured for the U.S. military by numerous corporations including Northrop, Lockheed, Beechcraft (now Raytheon), and the Globe Company (Goebel 2012). In 1955, Radioplane modified an early drone model to include a series of film cameras. The U.S. Army introduced these drones in 1959 and would utilize the technology throughout the Cold War for reconnaissance (Newcome 2004: 59). Although no open conflict ever occurred between the United States and the Soviet Union, reconnaissance flights were a fairly common, but dangerous mission. From 1946-1990, 23 aircraft and 179 servicemen were lost during reconnaissance flights related to Cold War operations (Newcome 2004: 71). Although most losses from such missions were kept quiet by both the U.S. and the U.S.S.R., the danger of the missions combined with the political turmoil experienced when airmen were captured prompted the U.S. Air Force to embark on a number of "surveillance drone" programs with companies like Radioplane, Northrop and others to produce nearly 1,500 drones. (Newcome 2004: 72-73).

The Cold War led to further developments in drone technology due to the "Space Race" between the U.S. and the Soviet Union and the nuclear arms race. Both of these events further increased the U.S. military's demand for new technology. Drones were seen as potentially useful in weapons testing after nuclear tests by pilots resulted in radiation-related illness and fatalities. The increased desire for UAV technology can again be seen in the number of private companies contracted by the military to produce UAVs. From 1946 through the 1960s, the Navy, Army, and Air Force contracted with Ryan Aeronautical, Beech, Curtiss, McDonnell, Globe, Martin, Radioplane, Northrop, Vought, and Lockheed to produce UAVs (Pasch 2010).

These investments contributed to even more advanced UAV technologies in the 1980s and 1990s. As a result of increased defense spending during the Reagan Administration, "microelectromechanical" system sensors (MEMS), mini global

positioning systems (GPS), and micro electronics became well developed, allowing for more advanced UAVs. Military operations abroad in Grenada, Lebanon, and Libya increased demand for inexpensive, unmanned, reconnaissance, and battle damage assessment (BDA) capabilities for field commanders. This demand, combined with the enhanced technology of the period led to further contracting between the U.S. military and drone manufacturers and yielded drones still in use today. In 1985, for example, the Secretary of the Navy ordered an expedited acquisition of UAVs for fleet operations. The RQ-2A Pioneer was introduced during this period and would operate in the Persian Gulf, Bosnia, Yugoslavia, and Somalia (U.S. Navy 2009).

Following the end of the Cold War and the collapse of the Soviet Union in the early 1990s, many of the existing Big Players in the private defense industry consolidated. Lockheed and Martin merged to become Lockheed Martin. Northrop and Grumman also merged. Lockheed Martin acquired British Aerospace and other smaller firms. Northrop Grumman and Boeing also made a series of acquisitions during the period. Mergers and acquisitions allowed the players in the defense industry to enhance their production capabilities while also increasing their political influence. A larger corporation with multiple locations meant that multiple agents in Congress would have a vested interest in growing the industry so as to bring money and employment to their districts. Hartung (2011: 20-21) describes how the acquisitions of British Aerospace and McDonnell Douglas by Lockheed Martin and Boeing were able to garner important political support for the companies:

Boeing was able to beef up [political ties by acquiring] McDonnell Douglas....British Aerospace came to join the Lockheed Martin Team. This gave Lockheed Martin a leg up

in persuading Britain to weigh in on its behalf. It is one thing to have a given state or senator in one's corner. It is quite another to have a sovereign state and longtime U.S. ally like the United Kingdom ready to go to bat for you.

The first Gulf War would prove a critical point in the development of the UAV

industry. According to a May 1991 report from the Navy, "at least one UAV was

airborne at all times during Desert Storm" (quoted in Frontline 2013). The conflict saw

522 separate drone launches and over 1,600 hours of flying time. During operations in

Iraq, the military, seeing the effectiveness of the UAVs, contracted familiar companies in

the industry to create and manufacture new drones. A report from the Oversight and

Investigations Subcommittee and Committee on Armed Services in 1993 stated the

impact of the technologies in the field:

[U]nmanned aerial vehicles (UAVs) provided substantial imagery support to Marine, Army, and Navy units during Operation Desert Storm. They were so good many more could have been used....These systems were employed for battlefield damage assessment...targeting...and surveillance missions, particularly in high-threat airspace....We could have used three times as many as we had. The Army took its solitary set of UAVs into the war and is now looking for many more. In one instance, Iraqi troops actually attempted to surrender to a UAV loitering over their position (quoted in Frontline 2013).

The performance of drones in Desert Storm was sufficient to once again increase

the demand for the technology. This general increase in demand was compounded,

however, by two broader changes in the military.

First, following the collapse of the Soviet Union, the U.S. military began to "transform itself in response to new threats" (Lyons 2004: 27). In prior decades, from before WWI through the Cold War, the U.S. had built up its arsenal to defend against the increases in its adversaries' weapon stocks. With the collapse of the Soviet Union in the early 1990s, however, the military lacked a clear antagonist and instead began to engage a "collection of asymmetric threats" (*Ibid*). While amassing weapons systems like bombers and tanks may have served to defeat the enemies of previous wars, such technology became impractical to build and deploy against multiple, smaller threats. Second, it was during this period that the U.S. Armed Forces sought to transform their role from one of traditional war fighters to one of "peacekeepers" (*Ibid*). Throughout the 1990s, the military undertook humanitarian operations in Somalia, Bosnia and Herzegovina, Rwanda, Kosovo, and elsewhere. Unlike previous military engagements, the goal was often not to force surrender of an enemy of the state, but to bring an end to conflict and provide humanitarian assistance (Seybolt 2008). Again, the use of traditional full-scale military operations and equipment proved difficult and often impractical with modern drone technologies seen as the main alternative.

It was this changing landscape of international politics and U.S. military operations which laid the final groundwork for the widespread use of drones in the new millennium. The shift from full-scale military operations to smaller engagements, combined with the success of prior drone use and new technological achievements would set the stage for drones to be used on a never before seen scale.

4 The Post-9/11 Expansion of Drones and Entanglement

The increased demand for drone technology following the Gulf conflict was augmented substantially by the post-9/11 conflicts in Afghanistan and Iraq. These conflicts, coupled with the broader Global War on Terror, created an opening for the expanded use of drones on an unprecedented scale. This is evident in the Department of Defense spending on Unmanned Aerial Systems (UAS) shown in Figure 1.



Figure 1: DOD Spending on UA: 1995–2013 (in millions)¹

As illustrated by Figure 1, spending on drones increased significantly in the post-9/11 period from \$363 million in 2001 (\$4.77 million in 2013 dollars) to \$2.9 billion in 2013. Further, between 2002 and 2010, the DOD's UAS inventory increased fortyfold (Gertler 2012: i). While only five percent of military aircraft were unmanned in 2005, by 2012 UAVs accounted for one third of *all* military aircraft (Gertler 2012: 9). An inventory of the DOD's unmanned aerial systems in 2003 counted only 163 UAS in use. The 2012 inventory, meanwhile, reported the DOD maintained a total of 7,454 unmanned

¹ Source: DOD, UAS Roadmap 2005-2030 (p.37) and DOD FY2009-2034 Unmanned Systems Integrated Roadmap (p. 4). All figured adjusted to 2013 dollars.

aerial systems—an increase of over 4,400 percent in less than a decade (Gertler 2012: 8).² The political economy framework can shed light on how this dramatic increase in drones came about.

One key driver was increased demand for drones by the military for combat purposes. Just as humanitarian missions in the 1990s had shown the expense and ineffectiveness of traditional military engagements, the multiple post-9/11 conflicts provided further evidence that "traditional" combat operations were no longer suitable. David H. Lyon (2004: pp. 27-28), chief of Advanced Munitions Concepts with the U.S. Army Research Laboratory, described the changing demands of military engagements in response to the threats posed by the U.S.'s new enemies and suggested a trajectory for military equipment and engagements.

[T]he U.S. Army can no longer afford the time and resources required to mass heavy vehicles prior to execution of an attack, as past doctrine dictated. Instead, the U.S. Army will consist of a collection of smaller, rapidly transportable units with the ultimate goal of deploying anywhere on the globe within 72 hours....Even with highly advanced technologies...the only way to achieve success...is to possess an unprecedented level of battlefield situational awareness. This will allow the U.S. Army systems to locate and engage targets at distances that far exceed enemy capabilities, such that significant attrition will cripple the enemy's ability to fight effectively, long before closing within traditional engagement ranges.

Drones provided all these modern features. They were smaller, well-equipped, and easier to deploy than traditional equipment. The data-gathering abilities of UAVs allowed for enhanced situational awareness. Their attack capabilities provided the military a chance to engage targets much faster than mobilizing ground units. Further, drones were able to enter and remain in environments which were not conducive to

² These inventories do not include small UAVs, micro UAVs (small enough to be "manportable"), or "lighter-than-air" (inflatable) platforms.

manned aircraft and remain in flight for over 24 hours. They are also able to perform additional functions such as surveying land and measuring cellular, radio, and other technological coverage over a variety of terrains (Unmanned Aerial Vehicle Systems Association 2013).

In addition to the changing organization of military engagements, the type of conflict in Iraq and Afghanistan gave the UAV technology additional appeal. While previous conflicts (e.g. WWI, WWII, and Vietnam) were often fought out in the open or on the battlefield, these military operations engaged enemies across a variety of terrains. As opposed to fighting an organized unit of enemy troops, the U.S. military was often seeking a particular individual or small group without precise knowledge of their location. Given the geographic realities of these battles, UAVs, some able to remain in flight for forty hours or more, could track and attack targets more effectively than traditional combat methods.

Drones also provided a mechanism for engaging in conflict while limiting U.S. soldier fatalities. Soldier deaths in post-9/11 conflicts, particularly in Iraq, were met with sharp criticism from citizens, the popular press, and both sides of the political aisle (Kuhnhenn 2008, NBC News 2007). By using drones, many missions would no longer require as many U.S. soldiers in the field, decreasing the number of casualties.

Although the military demand created by Afghanistan, Iraq and the broader War on Terror was central to the expansion of the drone industry, it was not the sole driver of the industry's growth. Established Big Players in the defense industry, like Lockheed Martin, Northrop Grumman, Boeing, General Dynamics, and General Atomics saw U.S.

military engagements as opportunities to earn significant profits. It is not difficult to see why. These engagements would mean more defense spending, including on UAV technologies. Consider that by 2012, world governments were spending more than \$6.6 billion on UAVs annually. Profits are expected to increase to \$11.4 billion annually over the next decade for a world market to more than \$89 billion a year (PR Newswire 2012). According to the GAO, the 2012 DOD acquisition plan for UAVs will require extensive payments to drone manufacturers. More than \$34.9 billion will be needed to purchase the desired number of drones from manufacturers (Gertler 2012: 33). In addition to the profits from the manufacture of new drones, the industry may expect additional revenues from drone repairs and modifications. Given these potential profits, coupled with the increased demand from the military, the result, as predicted by the political economy model, was intense rent-seeking by those in the defense industry.

Consider that top drone manufacturers have spent millions on lobbying expenditures since the Global War on Terror began in 2001. For example, leading up to and in the year following the U.S. invasion of Afghanistan, General Atomics, the maker of the Reaper, Warrior, Avenger, and Predator drones, significantly increased its lobbying expenditures. Between 2000 and 2002, General Atomics' lobbying increased by a compound rate of over 49 percent per year. During the same period, Northrop Grumman increased their expenditures at a compound rate of nearly 27 percent per year (CRP 2013).

Lobbying by the Big Players in the drone industry—Lockheed Martin, Northrop, Boeing, General Dynamics, and General Atomics—increased again over the 2008

election cycle. Between 2006 and 2008, the Big Players in the drone industry increased their lobbying at a compound rate of 17.41 percent per year compared to nine percent for total lobbying in the U.S. (CRP 2013). Lockheed, General Dynamics and Boeing increased their lobbying expenditures at a compound rate of 22, 12, and 35 percent, respectively, between 2006 and 2008 (CRP 2013). While overall lobbying in the U.S. decreased at a compound rate of 4 percent per year between 2010 and 2012, lobbying for the drone industry fell less than one percent (CRP 2013). Northrop Grumman, Lockheed Martin, and General Atomics, however, continued to *increase* their lobbying expenditures at compound rates of three, six, and eight percent, respectively, per year (CRP 2013).

In addition to increasing the overall level of lobbying spending Lockheed Martin, Northrop Grumman, Boeing, General Dynamics and General Atomics each retained multiple lobbying firms, many of which employed former government workers. In 2012, for example, 72.8 percent of lobbyists employed by Lockheed had formerly held a government job. Similarly, 67.3 percent of Northrop lobbyists, 71.9 percent of those retained by General Dynamics, 76.5 percent of Boeing's, and 72.2 percent of General Atomics' lobbyists were former government employees (CRP 2013). This "revolving door" between government and lobbying firms lowered the cost of traversing the Congressional landscape to have maximum influence on key decision makers.

Several of the lobbying firms retained by the top drone manufacturers were connected directly to former Congressmen. Former Congressman David Hobson, for

example, was a House representative from 1991-2009 and a member of the House Appropriations Committee and Subcommittee on Defense. Following his political tenure, Hobson founded his own lobbying firm, now retained by Lockheed Martin (CRP 2013). Jack Edwards, a long time member of the House, began working as a consultant for a D.C. lobbyist following his term. In 2012 alone, the firm received more than half a million dollars for its services from Lockheed, Northrop, and General Dynamics (CRP 2013). Former Congressman Sonny Callahan likewise began working as a lobbyist following his stint at the Capitol and also started his own firm. The two firms collectively received more than \$1.33 million in funds from drone makers during Callahan's tenure (CRP 2013). Former House member John Breaux and former Senate majority leader Trent Lott began their own bipartisan lobbying firm, Breaux Lott Leadership Group in 2008 and are also retained by the industry's Big Players, like Lockheed Martin (CRP 2013).

The attempt by the major players in the drone market to influence the activities of the government may also be seen in campaign contributions. While lobbying is generally conducted in an attempt to advance or hinder particular legislation, by contributing to a campaign, drone manufacturers look to curry the favor of politicians and influence the politics surrounding their industry more generally. Just as lobbying expenditures for the industry increased around elections and other major events like the invasions of Iraq and Afghanistan, so too did campaign contributions from the top drone manufacturers.

Between 2000 and 2002, total campaign contributions from Lockheed, Northrop, Boeing, General Atomics, and General Dynamics increased by over 13 percent to more

than \$10 million (CRP 2013). The elections of 2008 saw an increase in contributions of over 28 percent from 2006. In the elections of November, 2012, Lockheed, Northrop, Boeing, General Dynamics, and General Atomics increased their campaign spending to over \$11.9 million—an increase of more than 104 percent from the year before—while Northrop Grumman increased its contributions at a much faster rate—nearly 200 percent between 2010 and 2012 (CRP 2013). These campaign funds provided monetary incentives for members of Congress to be favorable toward spending on defense, including on expanding spending on UAV technologies.

In 2007, Congress passed the Fiscal 2007 Authorization Act, legislation that would all but guarantee the mass expansion of the industry. Among other requirements, the Act required the Secretary of Defense to

the Act required the Secretary of Defense to

[d] evelop a policy to be applicable throughout the Department of Defense, on research, development, test and evaluation, procurement and operation of unmanned systems....[The policy must include] a preference for unmanned systems in acquisition programs for new systems, including a requirement under any such program for the development of a manned system for a certification that an unmanned system is incapable of meeting program requirements (John Warner National Defense Authorization Act for 2007, section 941, emphasis added).

Thus, Congress drastically altered the way in which new systems were to be

designed. Instead of looking to develop unmanned systems to serve the same tasks as existing manned systems, unmanned systems would now be the assumed *starting point* for new technologies. Unmanned systems were to be developed for tasks unless some other need required new systems to be manned. The impact on the drone industry was drastic and rapid. By 2010, spending on drones increased to \$3.3 billion (Gertler 2012: i). By 2011, the DOD had increased their spending on drones to \$4 billion annually (Brinkerhoff 2011, Waldman 2013). As a result of the expansion, all the major players—

drone manufactures, the military, and Congress-reaped substantial benefits.

Perhaps the best demonstration of the impact the drone industry has had on Congress occurred in 2009 with the formation of the Congressional Unmanned Systems Caucus (CUSC). The CUSC is a group of approximately fifty members of the House of Representatives dedicated to expanding the drone industry:³

[The Unmanned Systems Caucus seeks to] educate members of the Congress and public on the strategic, tactical, and scientific value of unmanned systems; actively support further development and acquisition of more systems, and to more effectively engage the civilian aviation community....We acknowledge the overwhelming value of these systems to defense, intelligence, homeland security, law enforcement, and the scientific communities....We recognize the urgent need to rapidly deploy more Unmanned Systems in support of ongoing civil, military, and law enforcement operations....We work with the military...and other stakeholders...[to] support our world-class industrial base that engineers, develops, manufactures, and tests unmanned systems creating thousands of American jobs....We support policies and budgets that promote a larger, more robust national security unmanned system capability (CUSC 2013).

It is not difficult to understand why the representatives in the CUSC desire to

expand the industry. All members of the caucus come from a state with some connection

to drone manufacturers, meaning they have a vested interest in expanding the industry to

generate benefits for their constituents.

³ A similar caucus was created in the Senate in 2012 with a mission to "educate Senators and staff on the capabilities and concerns of UAS and to work closely together to best shape the UAS policymaking process (Harder and Heisten 2012). Just as the House UAS caucus contains members with vested interests in the continued expansion of the drone industry, so too does the Senate caucus. Also like the House caucus, the group in the Senate maintains close ties to the AUVSI and drone manufacturers. Michael Toscano, President of the AUVSI stated the Senate caucus would "enable AUVSI to work with the Senate and stakeholders on the important issues that face the unmanned systems community as the expanded use of the technology transitions to the civil and commercial markets....It is our hope to establish the same open dialogue with the Senate caucus as we have for the past three years with the House Unmanned Systems Caucus" (quoted in Aero News Network 2012).

Congressman Buck McKeon, one of the caucus' co-chairs, for example, represents the district in which Northrop Grumman manufactures its Global Hawk drones. His representative state, California, is poised to gain immensely from increased drone production. A 2013 report by the Association for Unmanned Aerial Vehicle Systems International (AUVSI), the industry's largest lobbyist, found that enhanced drone production would bring more revenue to California than any other state (AUVSI 2013: 3). Henry Cuellar, the caucus' second co-chair, represents a district just outside of San Antonio, the home of branches of Lockheed Martin, Northrop Grumman, and Boeing. The same AUVSI report found that Texas is poised to reap some of the most extensive benefits from the expansion of the drone industry and is third in potential revenue behind California and Washington state (AUVSI 2013: 3). Other members of the caucus have similar connections and hail from states which will also gain significantly from further drone production. More than half the members of the CUSC represent the ten states which the drone industry projects will see the most gains in terms of jobs and additional revenue with the increase in UAV production (AUVSI 2013: 3).

These relationships illustrate the logic of public choice, that the policies surrounding drones (like policy in general) are driven by the incentives facing policymakers. The Big Players in the market and the polity (i.e. the military, elected officials, and drone manufacturers) utilize a variety of means, including lobbying, campaign finance, political clout, and other pressures to influence the other groups and reap various benefits for their members. The central issue is whether these narrow interests align with broader notions of the public interest as it relates to defense.

5 Conclusion

The public controversy surrounding the use of drones by the U.S. government is a recent phenomenon. But as our analysis shows, the UAV technology itself is anything but new. Instead, there is long history of linkages between the U.S. government and private producers of these technologies. Our central purpose has been to document this history. In itself, this has value purely as an exercise in U.S. military history and political economy. But beyond this, the political economy of drones has important implications for current policy debates.

The current controversy over drones is multifaceted. On the one hand there is debate about the international use of drones. These controversies focus on the effectiveness of drones in targeted killings, the legality of drone use in the context of state sovereignty, and the ethics of drone use due to the possibility of collateral damage. At the same time, there is ongoing debate about the domestic use of drones related to privacy and civil liberties. Regardless of how these debates ultimately unfold, the creation of new policies surrounding the use and manufacture of UAVs is certain. It is in this policy arena where our analysis has the potential to illuminate and inform.

One of the central insights from public choice is that policy is not designed in a vacuum. Instead, public choice scholars emphasize that policy is the outcome of a process influenced by several key categories of actors: individual voters, interest groups, bureaucrats, and elected officials. Further, this policy process unfolds in the context of existing relationships, networks, and power distributions between the players based on past interactions and policies. Our analysis sheds light on both the historical context and current key parties relevant to UAV technologies. This is particularly relevant for policy

regarding drones. Given the existing entanglements and ability of Big Players to influence outcomes, first-best policies may be impossible or at best exceedingly difficult to obtain. While Big Players don't have complete influence over policy, they do have the ability to significantly alter outcomes, calling into question whether drone-related policies align with the interests of U.S. citizens more broadly.

Ignoring the political realities highlighted in our analysis may result in ineffective, or, worse, damaging policies. For example, given the current controversies over the domestic use of drones, several states have passed bans on the use of UAV technology in their airspace. While this might contribute to the protection of privacy and civil liberties on some margins, such blanket laws also run the risk of undermining private innovation that might yield widespread economic benefits for U.S. citizens (see Dourado 2013).

Lastly, our work sets forth a significant challenge to those undertaking the construction of drone policy. How does one design policies which constrain the narrow interests of those involved in the drone industry while maintaining the potential benefits offered by UAV technologies? Given the entrenched entanglements discussed throughout this chapter, this is no easy task. Such issues, however, must occupy the foreground of any policy discussion if we are to avoid perverse outcomes and obtain the best possible policies.
DRONES: PUBLIC INTEREST, PUBLIC CHOICE, AND THE EXPANSION OF UNMANNED AERIAL VEHICLES

1 Introduction

Unmanned aerial vehicles (UAVs), or "drones" have been described as the future of modern warfare. Observing one UAV on an aircraft carrier in 2013, Navy Secretary Ray Mabus stated, "It's not often that you get to see into the future, but that's what we got to see today" (quoted in Lubold and Reed 2013). Air Force Col. Eric Mathewson described drones as, "a revolution of military affairs" (quoted in Pappalardo 2010). In 2013, former Defense Secretary Donald Rumsfeld stated he expects drones to play a substantial role in the continued War on Terror stating, "there is clearly going to be a use for drones; there has been in the past and there will be in the future" (quoted in Sherfinksi 2013).

Although drone use is evolving on a daily basis, it is clear that UAVs are a core component in U.S. military operations. Indeed, drones have been used by the U.S. military in some capacity since WWII and UAVs were used extensively for surveillance missions abroad throughout the Cold War.⁴ But the start of the War on Terror in 2001 launched drone use to new levels. In addition to reconnaissance, drones are now used as a means of conducting offensive strikes (Sifton 2012). Since 2008, the U.S. has conducted more than 1,000 strikes in Afghanistan. Between 2008 and 2012, 48 strikes occurred in

⁴ See Hall and Coyne 2014, for a more complete analysis of the emergence of drones and their historical use by the U.S. military.

Iraq. As the U.S. prepares to redeploy some kind of force to the country in 2014, drones have reportedly been providing intelligence (see Entous and Barnes 2014). Over 140 strikes were launched in Libya. Four hundred drone attacks have occurred in Pakistan, more than 100 in Yemen, and at least one in the Philippines (Kreps and Zenko 2014: 71). As many as nine strikes have occurred in Somalia (Rogers 2012).

The expanded use of UAV technology is also reflected in the number of drones and UAV platforms employed by the U.S. military. In 2000, for example, the DOD employed approximately 50 drones and supported five distinct UAV platforms. By 2012, 15 unique platforms and over 7,000 drones were supported—a 140-fold increase in twelve years (Bone and Bolkcom 2003: i, Gertler 2012: 8, Tomiuc 2012). UAVs comprised a mere five percent of U.S. military aircraft in 2005; by 2012, they accounted for 41 percent (Gertler 2012: 9). Plans to further expand the U.S. drone arsenal are scheduled through 2038 (U.S. DOD 2013: 72).

Academic scholars and policy makers have assumed, often implicitly, that the increased use of UAV technology and decisions regarding drones have been, and will be, made in the "public interest." That is, it is assumed that those who design and implement drone policy set aside private incentives and construct policies to maximize the production of national defense and security. The purpose of this chapter is to explore a number of conjectures that follow from this assumed public interest and to examine the robustness of these predictions. To the extent these claims are not supported empirically, this work seeks to offer an alternative explanation to reconcile observed policy outcomes and the public interest ideal.

This chapter contributes to two strands of literature. The first is the small, but growing body of work on UAVs. Existing scholarship on UAVs tends to focus on the legal or ethical implications of drone use (American Civil Liberties Union 2011, Arkin 2010, Dipert 2010, Jenks 2010, Sharkey 2010, Singer 2009, Strawser 2010). Others provide a historical account of UAV use, examine various technical aspects of drones, or look to provide a cost and benefit analysis of the technology's use in combat or other missions (M. Boyle 2013, Byman 2013, Cronin 2013, Kreps and Zenko 2014, Lyon 2004, Miller 2012, Newcome 2004). Throughout this literature, it is largely assumed that those influencing the research, development, acquisition, and other policies surrounding drones, make their decisions in the public interest. This chapter contributes to this literature by offering the first examination of the robustness of the assumption of publically interested drone policy. My analysis focuses specifically on the period following September 11, 2001, since this is when one observes the most dramatic increase in the use of drones (see Hall and Coyne 2014). I limit my focus to the use of UAVs by the U.S., as the various agencies of the U.S. government maintain and employ more UAVs than any other group.

Second, this work contributes to the larger literature on defense and peace economics (see Anderson and Carter 2007, 2009, Arrrow 1995, Boulding 1945, 1978, Brauer and Caruso 2012, Brauer and Dunne 2012, Coyne 2008, 2005, Coyne and Cowen 2005, Coyne and Pellillo 2011, Hartley and Sandler 1995, Hirshleifer 2001, Isard 1992, Poast 2006, Sandler and Hartley 1995, Smith 2009). In particular, this work contributes to the scholarship on the defense industrial base. This literature explores the relationship

between the government's demand for, and expenditures on, military goods and services, and the private producers of those goods and services (see Dunne 1995, Hartley and Sandler 1995, O'Hanlon 2011, Watts 2008). Particular focus is placed on the interplay between the two categories of actors to understand how government influences the members of the defense industrial base and vice versa.

My analysis contributes to this literature by offering an analysis of UAVs in the Global War on Terror with a particular focus on and the broader institutional structure under which drone policy is formed. I offer insight into the interplay between the relevant actors in government and in private industry and how their interaction generates outcomes that are often at odds with what the public interest would predict.

The rest of this chapter proceeds as follows. Section 2 lays out the public interest framework as assumed by the existing scholarship on UAVs. This section identifies what one would expect to observe with respect to drone policy given this assumption. Section 3 provides empirical evidence related to the conjectures derived in section 2 in order to examine their accuracy and robustness. Section 4 offers a discussion of the alignment of expected and observed policy and provides and alternative analytical structure for examining drone policy. Section 4.1 discusses the incentives facing private interests in the drone industry. Section 4.2 describes the incentives facing political actors and their means for influencing others. Taken together, these sections provide an in-depth analysis of UAV policy. Section 5 concludes with implications.

2 Public Interest and Implications for Drone Policy

When discussing the construction of defense policy in general, it is assumed by the literature that those involved set aside their own goals and work to serve the greater "public good." Benevolent agents are assumed to be motivated by some larger social welfare function which includes the provision of national defense. UAVs represent a component of this national defense variable. Policymakers, acting as a collective unit, allocate resources so as to maximize the value of national defense (i.e. provide the best protection for the U.S. citizenry) and the larger social welfare function. These actions are supposedly reinforced by appropriate feedback mechanisms. Namely, these public actors are motivated to please their "employers"—the general public (see Tullock 2008). It follows that the creation or alteration of UAV policy benefits society as a whole rather than a subset of particular actors and directs resources to their highest-valued use.

In the literature on UAVs, this framework of public interest is, at least implicitly, taken as given. Even in works where the ultimate use of drones is criticized, it is assumed that political actors look to provide the best possible defense for U.S. citizens and have chosen drones as the means to achieve these goals. Abizaid and Brooks (2013), for example, while offering a number of suggestions to improve drone policy, ultimately assume policymakers will look to fulfill the public's goals. Discussing future UAV developments, they state (15), "[policy should be] geared toward advancing US national security interests in a manner consistent with [the public's] values." Similar assumptions of public interest are observed elsewhere, "[P]rotecting vital public interests and promoting innovations that stand to substantially benefit society requires…laws, policies, and regulatory frameworks" (Drones and Aerial Robotics Conference: 1). From this

assumption of public interest, one may derive two general conjectures, each with several subsidiary conjecture, regarding what drone policy should look like in practice.

Conjecture 1: Defense expenditures on the production of UAV technology are allocated to maximize defense and security for U.S. citizens

According to the public interest view, policies are constructed to serve the interests of the broader public. With regard to drones, this implies that UAVs are produced because they provide the best possible defense and security for U.S. citizens. This conjecture yields two sub-conjectures regarding the costs of drone production.

Conjecture 1.1: Producing drones is the most cost effective means, relative to known alternatives, of achieving U.S. security objectives.

If drones are produced in the public interest, this implies that drones provide the same, or more, "security output" at a lower price. This indicates that, as the U.S. has shifted operations from manned aircraft to drones, we would expect drones to provide the same, or greater, outputs as manned aircraft (surveillance, targeted strikes, etc.) for a lower price. This assumption of drones' cost efficacy is observed throughout the existing literature on UAVs. For example, one author notes that, "Drones reduce the dollar cost of using lethal force...[they] are a bargain compared with the available alternatives" (Brooks 2012), while another indicates that, "Drones are substantially cheaper than traditional ground forces" (Francis 2013), and yet another author argues for the desirability of UAV technology because, "They're cheap" compared to alternative technologies (Ratnesar 2013).

The assumption of public interest in the context of cost reduction has further implications. In the event the cost to using UAVs was greater than that of an equally effective alternative, or the cost of UAVs was unclear, benevolent government actors would reallocate resources toward producing alternatives so as to provide the best possible defense at the lowest price.

Conjecture 1.2 Drones provide a technically and operationally superior means of defense relative to alternative technologies.

If drones are produced within the public interest, this implies that drones are not only the most cost-effective option, but are also generally technologically superior to the alternatives for providing defense. This assumption is seen throughout the literature. Zuckerman (2013), for example, demonstrates this assumption stating, "the drone is an incredibly effective tactical instrument."

Within this context of technological superiority, the public interest assumption has further implications. Specifically, if UAVs are operationally suspect, or generally less technologically effective than other forms of defense given their relative prices, than resources would be reallocated to other forms of defense as UAVs would not be providing the best possible defense for U.S. citizens.

Conjecture 2: Ineffective or counterproductive drone policies would be eliminated or modified

The public interest view implies that drones are the best tool for conducting U.S. foreign policy and carrying out U.S. objectives—most notably, goals related to the War on Terror and other issues of national security. From this three sub-conjectures emerge.

Conjecture 2.1 Drones are superior at detecting, targeting, and dismantling terrorists and other enemies better than manned aircraft or other means.

It is argued that drones have a superior ability to eliminate high-value terror

targets, thus crippling the leadership of terrorist groups which look to harm the interests

of U.S. citizens. These ideas are captured clearly in President Obama's (2013) speech on

drone policy which, among other things, noted that,

[Drones] are effective....Dozens of highly skilled al Qaeda commanders, trainers, bomb makers and operatives have been taken off the battlefield....[T]he primary alternative to [drones] would be the use of conventional military options....Conventional airpower or missiles are far less precise than drones.

Byman (2013: 33) echoes these statements arguing that, "Drones...undercut

terrorists' abilit[ies]....[T]raining on a large scale is nearly impossible when a drone strike could wipe out an entire group of new recruits....[T]he alternatives [other technology] are either too risky or unrealistic." The assumption of public interest within this context implies that if UAVs do not provide superior means of eliminating terrorist targets, we would expect to see a substitution away from drones toward other means of combating terrorism.

Conjecture 2.2 Drones are more effective at reducing collateral damage, minimizing civilian casualties, and at reducing potential harms to troops relative to alternative technologies.

If drone policy is constructed within the public interest, this implies that, along with possessing a superior ability to dismantle terrorist groups, UAVs are better able to reduce the number of civilian casualties and other damage compared to the alternatives. The minimization of collateral damage is important in this context for two reasons. First, casualty reduction is proposed to reduce instances of "blowback" (the unintended negative results of military action) which may harm U.S. citizens both domestically and abroad. Second, casualty reduction works to follow international laws which require governments to minimize civilian deaths. Failure to do so may result in sanctions which adversely impact the U.S. citizenry.

The idea that drones reduce casualties is readily observed. CIA Director John Brennan, for example, stated that drones maintain "surgical precision—the ability with laser-like focus to eliminate the cancerous tumor called an al Qa'ida, while limiting damage to the tissue around it" (Brennan 2012). Others have made similar claims stating that, "drones kill fewer civilians...than any other weapon" (Saletan 2013) and, "[drones are] actually the most human form of warfare" (Lewis 2013).

Though distinct from the issue of civilian casualties, the public interest assumption likewise implies drones are superior at reducing harm to military personnel. Placing fewer troops on the ground implies lower deployment costs, as well as lower costs on ex-post medical and other care, thus reducing the cost to U.S. citizens. Burris (2013) captures this idea plainly, "Drones spare American soldiers." If UAVs do not in fact reduce civilian and military casualties, we would expect to see other methods employed and a reevaluation of drone policy in order to fulfill the public interest.

Conjecture 2.3 Government officials responsible for constructing UAV policy will utilize the best information available to create, evaluate, and alter drone policies to maximize social welfare.

If drone policy is constructed within the public interest, we would expect those responsible for designing and implementing UAV programs to utilize the best information available. If those with a comparative advantage in defense and security activities offer information or advice suggesting an alternative or modification to UAV policy, we would expect to see these suggestions integrated into UAV policy. In this case, we would expect policymakers to pay particular attention to reports and opinions from the U.S. military and other counterterrorism experts. These individuals have a comparative advantage in defense and security issues and, therefore, are most qualified to offer information regarding the aforementioned issues of casualties, technical efficiency, and operational effectiveness.

3 Data and Evaluation of the Public Interest Conjectures

Whether the conjectures associated with the public interest assumption hold is an empirical question. Unfortunately, given that drone use has only recently increased, and that many key aspects are classified, "large-n data" is unavailable for standard statistical tests. In the absence of appropriate quantitative data I draw on data from a variety of sources to empirically analyze the public interest conjectures. These data sources include: official statements and documentation from the U.S. government, works from reputable media outlets, and academic papers. While this approach is less formal than standard statistical analysis, it is appropriate given the importance of the topic and the limited quantitative data. Moreover, the goal is not to make specific point predictions regarding UAVs, but rather to draw on available sources to discern broad patterns in the context of the conjectures established above. If drone policy is constructed solely to serve the public interest, we should expect to see strong evidence supporting each of the above conjectures, as well as general agreement among experts on their implications. Evidence contrary to these conjectures, or substantial disagreement regarding these claims would

indicate that motivations outside the public interest are impacting the creation,

implementation, and ultimate use of drones.

Conjecture 1: Defense expenditures on the production of UAV technology are allocated to maximize defense and security for U.S. citizens

Conjecture 1.1: Producing drones is the most cost effective means, relative to known alternatives, of achieving U.S. security objectives.

The available evidence casts doubt on the suggested cost efficacy of drones. At best, drones appear to provide a minimal cost advantage to comparable manned aircraft. At worst, UAVs provide no more security than manned aircraft and are significantly more expensive. Taken together, this suggests the public interest framework may be insufficient in explaining the current utilization of drones by the U.S. government. Using declassified reports from the DOD, A. Boyle (2012) found that UAVs provided only a slight cost advantage over manned systems. Abizaid and Brooks (2014: 22) state that, "UAVs are not inherently cheaper than manned aircraft." Bone and Bolkom (2003: i) caution that the differences between the projected and actual price of many UAVs is substantial. The accident rate for drones is in some cases 100 times higher than those of comparable manned aircraft. As of July 2010, the Air Force had identified 79 drone accidents costing at least \$1 million each (Zucchino 2010). When taking these higher fail rates into account, the overall cost-effectiveness of UAVs is far less clear. Others note the "savings" created by UAVs may be partially or completely offset by the need for multiple "ground pilots" and that while per unit costs may appear smaller, in the aggregate, total acquisition costs rival those of other weapon systems (Bone and Bolkcom.: 12).

These doubts regarding cost minimization are illustrated by examining cases of two UAV platforms—the Block 30 Global Hawk and the Fire Scout. The Global Hawk, for example, was initially priced at \$35 million a unit, well below the cost of a comparable manned aircraft. The actual cost to manufacture the UAV, however, was more than \$200 million per unit, significantly higher than comparable alternatives (Roston 2013). The Fire Scout, another UAS, likewise saw costs increases so significant, law required Congress to be notified (Gertler 2012: 10). As of December, 2013, the average procurement cost of the Fire Scout had risen some 71.5 percent (Carey 2014).

Conjecture 1.2 Drones provide a technically and operationally superior means of defense relative to alternative technologies.

Just as the available evidence regarding the cost efficacy of drones is conflicted and may contradict the implications of publically interested policy, so too does the evidence regarding the operational aspects of UAVs. A variety of reports, including those by top military officials, indicate drones are operationally suspect. Gen. Mike Hostage, chief of the air service's Air Combat Command, for example, described the drones most frequently used in Iraq and Afghanistan, "useless in [many] environment[s]....[Drone are] not the force structure the nation needs or can afford" (quoted in Reed 2013). Dyke Weatherington, head of the DOD's UAS planning taskforce stated that issues with drones had complicated missions to the point that the technical effectiveness of UAVs had been greatly reduced (see Gertler 2012: 16).

These questions of operational efficacy are further illustrated by specific cases. For example, discussing the aforementioned Fire Scout UAV, a top Pentagon official stated the Fire Scouts performed poorly during tests and that they drones were "were not operationally realistic" (quoted in Friess 2011). In discussing the Global Hawk, another Pentagon official said bluntly, "the Block 30 is not operationally effective" (Sia and Cohen 2013). Incidents regarding the UAVs becoming disconnected from their human operators, straying into restricted airspace, and crashes have been reported, raising additional concerns regarding the UAVs operations (Pringle 2012).

While the data above demonstrates operational issues with UAVs, determining whether or not drones provide the "best possible defense" requires a comparison to the alternatives—namely, manned aircraft. Again, it is unclear UAVs are generally superior to their manned counterparts. Discussing the ISR (intelligence, surveillance, and reconnaissance) capabilities of manned and unmanned aircraft, Army Lt. Col. James Cutting, the service's unmanned air systems division chief stated, "manned platforms are more effective than unmanned platforms" (quoted in Moorman 2009).

These questions of technical differences between manned and unmanned aircraft is again well-illustrated with a specific case. Consider, for example, the Global Hawk UAV. In 2012, when faced with the choice between the U-2 or the Global Hawk, the Air Force decided the manned U-2 was better suited for its future needs given forecasted budgets, the respective payloads of the two vehicles, technological abilities, and the difference between their mission success rates. This was in spite of the fact that the Global Hawk had been created in an effort to replace the U-2 (see Thompson 2014).

Taken together, this disagreement regarding the effectiveness of UAVs and evidence suggesting manned aircraft may be generally preferable than drones,

demonstrates that current drone procedures do not appear to completely align with the implications of the public interest assumption.

Conjecture 2: Ineffective or counterproductive drone policies would be eliminated or modified

Conjecture 2.1 Drones are superior at detecting, targeting, and dismantling terrorists and other enemies better than manned aircraft or other means.

Evidence suggests that, at a minimum, drone strikes are not intrinsically better than other means at targeting and eliminating terror threats. In the worst case, current UAV policy has led to an increase in terrorist group membership and an expansion of terror activity.

One study of over 250 terrorist groups found that most terrorist groups cease operations when group members decide to join in the political process (43 percent) or *local* law enforcement dismantle key members of the group (40 percent). Only seven percent of terrorist groups ended through military force, suggesting that drones, and military strikes in general, are not the best method for eliminating terrorists (Jones and Libicki 2008: 18-19).

When drone strikes are used, it is unclear that drones are superior to other methods at eliminating the desired "high level" fighters. Since 2008, government sources show that drone strikes have killed approximately twelve times more low-level fighters than mid-to-high level al Qa'ida and Taliban leaders (Entous 2010). Another study found that the number of high-level targets killed as a percentage of total casualties was a mere two percent (International Human Rights and Conflict Resolution Clinic at Stanford Law School and Global Justice Clinic at NYU School of Law 2012: vii). Counter to the publically interested idea of decreasing the threat of terrorism, there is strong evidence that drone strikes are contributing to an *increase* in recruitment to terrorist organizations and terrorist activity. One study found that not only had the number of terrorist attacks increased between 2007 and 2013, but the number of al Qa'ida affiliated jihadist groups increased by 58 percent between 2010 and 2013 (Jones 2014: x). Reporting on drone strikes and the reduction of terrorist groups, Jones (2014: 60) found, "there is mixed evidence, at best, that drone strikes…are effective in defeating terrorist groups." Similarly, Smith and Walsh (2013) found that drone strikes were ineffective at degrading al Qa'ida and led to an increase in propaganda output. Kilcullen and Exum (2009) describe how drone strikes become a recruitment tool for terror organizations, "Every one [killed in a strike] represents an alienated family, a new desire for revenge, and more recruits for a militant movement that has grown exponentially even as drone strikes have increased."

Conjecture 2.2 Drones are more effective at reducing collateral damage, minimizing civilian casualties, and reducing harm to troops than other means. It is unclear that drones reduce civilian casualties or place fewer U.S. soldiers in

harm's way. One study of civilian casualties from drone strikes stated that, "[T]he dominant narrative about the use of drones...[that they are] surgically precise and effective...is false" (International Human Rights and Conflict Resolution Center 2012: v). Using classified military data on drone strikes and the civilian casualties they caused from 2010-2011, researchers with the Center for Naval Analyses found that drone strikes were *ten times more deadly* to Afghan civilians than strikes performed by manned aircraft. "When pilots flying jets were given clear directives...on civilian protection,"

they stated, "[manned fighters] were able to lower civilian casualty rates [compared to UAVs]" (quoted in Ackerman 2013). Moreover, the current official definitions of "militant" and "civilian" may bias official casualty estimates. A "militant" is officially defined as "all military-age males in a strike zone," meaning that the true identities and affiliations of many targets are never properly recorded (Wolverton 2013).

It is also unclear that drones provide a superior means of protecting U.S. military personnel from potential harms. In fact, using UAV technology may require *more* of a military presence in a given area as opposed to less. Zenko (2012), states that "drones require more boots on the ground" as the UAVs require a "ground pilot," a large number of surveillance analysts, maintenance personnel, and sensor operators. Further, it is unclear that drones spare soldiers from the psychological costs of combat. Studies from the Defense Department found that drone operators experience mental health issues at the same rate as conventional pilots (Dao 2013). An Air Force study of UAV operators found that nearly half reported "high operational stress." Nearly 25 percent displayed "clinical distress," anxiety, depression, or stress severe enough to impact job performance and family life. (Bumiller 2011).

The above analysis indicates the narrative regarding casualty reduction is at best unclear and raises doubts regarding the proposal that drones protect civilians or soldiers. At worst, drones not only increase the number of civilian casualties, but also increase the number of U.S. personnel in the field. In this case, current policy would be in direct conflict with the assumption of publically interested policy.

Conjecture 2.3 Government officials responsible for constructing UAV policy will utilize the best information available to create, evaluate, and alter drone policies to maximize social welfare.

In the case of drones and defense policies, we should expect policymakers seek

out those with a comparative advantage in military and counterterrorism experience and

to incorporate this feedback into their decision making. While it is impossible to know of

all the correspondence between the military and policymakers, there is evidence to

suggest that the advice of experts with regard to drones is being ignored. A variety of

former military and counterterrorism experts have pointed out a number of issues

regarding technical and allocative issues, and called for radical policy changes (see Friess

2011, Gertler 2012, Reed 2013, Zenko 2012, Zucchino 2010).

In 2013, Retired Gen. Stanley McChrystal, former Commander of U.S. Forces in

Afghanistan and former Director of the Joint Special Operations Command, offered the

following precaution regarding UAVs,

[Drones] didn't solve the problems [in Afghanistan]. The tactics that we developed...don't produce decisive effects....If we were to use our technological capabilities [drones] carelessly...then we should not be upset when someone responds with their equivalent, which is a suicide bomb in Central Park, because that's what they can respond with (quoted in Byers 2013).

Other experts have called for serious changes to drone policy. In 2014, a panel of

defense experts, led by Gen. John Abizaid, former commander of U.S. Central

Command, and Rosa Brooks, former counselor to the Undersecretary of Defense

expressed concern over using drones stating that reliance on drones, "rests on

questionable assumptions, and risks increasing instability," (28) and that "[drone] strikes

are [in]consistent with core rule of law norms" (36).

Despite these concerns, it does not appear that policymakers are heeding the advice of experts. In 2013, for example, the Air Force attempted to retire one line of UAVs, citing numerous reports of operational issues and extreme cost overruns. As opposed to discontinuing the UAV however, lawmakers instead *required* the Air Force to continue to use the drone and purchase two additional units despite the military's express desire to eliminate the program (see Sia and Cohen 2013). Drones continue to be used widely for surveillance and offensive operations and despite some efforts to increase transparency and make other changes to the way UAVs are used, these changes have either been blocked, or stagnated during the political process.⁵

4 Public Choice as an Alternative Framework

The above suggests that an alternative framework is required to resolve the discrepancies between the public interest ideal and the empirical evidence. How may one explain the continued expansion of UAVs despite evidence their use may not align with, or even contradict, the public interest?

The answer to this question requires an appreciation of the incentives facing a variety of actors within the UAV industry and an understanding of the broader institutional context in which these actors operate. As Buchanan (2003), Brennan and Buchanan (1985), Buchanan and Tullock (1962), Tullock (2005), Niskanen (1971), Olson (1965) and others have noted, the alternative to the public interest assumption is the "public choice" framework which emphasizes a symmetry of behavioral assumptions across contexts. The framework does not deny that individuals in politics may value the

⁵ For a discussion of these proposed changes, see Miller 2014.

wellbeing of others or, in the case of this chapter, the provision of "national defense" to some degree. However, they argue that individuals in positions to influence the trajectory of policy do not, as the public interest framework suggests, look solely to maximize some larger social welfare function. Instead, they posit that policymakers, just like private actors, respond to incentives created by political institutions in which they interact (see Buchanan 2003). Institutions, as the "rules of the game," determine the incentives faced by policymakers (North 1990). According to their model, public actors will pursue the public will only if their private interests overlap with the public interest. This insight leads to a critical observation—if the existing political structure through which drone policy is constructed does not appropriately align private and public interests, then one would expect to observe a wedge between witnessed policy outcomes and the outcomes one would expect in the case of pure public interest.

To explain the discrepancies and questions raised in section 3, I examine two core groups from the public choice model—special interest groups (namely, defense contractors) and politicians (Congress and other elected officials). As per the public choice model, it is assumed each group maximizes their respective payoffs subject to the constraint they face. In so doing, the groups look to influence the trajectory of, in this case, defense policy. It is further assumed that each groups possesses something the other group finds desirable (e.g. ability to make laws, access to monetary funds, etc.). Responding to their own incentives regarding UAVs, these groups look to influence one another. The individual goals of each group, combined with the desire to influence one

another, results in exchanges between the groups. Through these trades, the private interests of each group are enhanced.

4.1 Private Industry

Special interest groups, in this case private defense contractors, face a strong incentive to obtain government contracts. As profit maximizing entities, the firms responsible for designing and manufacturing UAVs face strong incentives to pursue policies which increase their revenues, even if these pursuits do not align with the public interest. In order to further their interests, defense contractors have two distinct means of influencing drone policy—lobbying and campaign contributions.

The industry has undertaken substantial lobbying efforts in the post 9/11 period, as well as contributed millions in campaign contributions to Congress. Taken together, these methods have been effective in influencing the trajectory of drone use by U.S. forces. The five greatest-contributing drone manufacturers (Lockheed Martin, Northrop Grumman, General Dynamics, General Atomics, and Boeing) expanded their lobbying efforts substantially with the start of the War on Terror in 2001. Between 2000 and 2002, for example, General Atomics and Northrop Grumman increased their lobbying expenditures by compound rates of over 49 and 27 percent, respectively, per year. As a point of comparison, total lobbying expenditures in the U.S. rose at a compound rate of six percent per year over the same period (Center for Responsive Politics (CRP) 2013). During the 2008 elections, these five firms increased their lobbying at a compound rate of 17 percent per year on average compared to a nine percent increase in total U.S. lobbying (*Ibid*). Figure 2 illustrates the combined lobbying expenditures of these firms between

2000 and 2012. Within this period, these five firms spent over \$600 million in lobbying efforts.



In addition to lobbying efforts, these five players contribute to a variety of political campaigns of *both* political parties. Between 2000 and 2002, for example, total campaign contributions from Lockheed, Northrop, Boeing, General Atomics and General Dynamics increased by over 13 percent to more than \$10 million (*Ibid*). Comparatively, the five industries which contribute the most to political campaigns decreased their total campaign contributions by 0.6 percent.⁷ By the elections of November, 2012, campaign

⁶ Source: Center for Responsive Politics 2013. All figures have been adjusted for inflation.

⁷ These industries comprise those firms involved in pharmaceuticals, insurance, electricity, business associations, and computers.

contributions from the top UAV manufacturers increased to over \$11.9 million—an increase of more than 104 percent from 2010. Firms like Northrop Grumman increased spending at a faster rate, upping contributions more than 200 percent between 2010 and 2012 (*Ibid*).

Though it is always difficult to make a direct causal link between lobbying expenditures and political outcomes, there is evidence that contractors have had substantial impact on the trajectory of drone policy, including those policies which the above evidence suggests do not align with pure public interest. One report by the Association for Unmanned Vehicle Systems International (AUVSI), the main lobbying organization for the industry, stated, "[We're the] go-to advocacy group for staffers and Members on Capitol Hill" (Toscano 2011). One example of the success of these efforts regards the reauthorization legislation for the Federal Aviation Administration (FAA). In a presentation by AUVSI, CEO Michael Toscano reported that, "The only changes made to the UAS sections of the House FAA bill were made at the request of AUVSI. Our suggestions were often taken word-for-word" (Ibid, emphasis added). This legislative change is one clear example of policy which may not align with the public interest. The passage of the act requires the FAA to integrate drones into domestic airspace by 2015, even though such an integration has raised significant concerns over issues of privacy, abuse by domestic law enforcement, and lack of appropriate legal frameworks. Each of these issues may undermine the safety and liberty of the citizenry.^{8,9}

⁸ See Murray 2012 regarding public opinion on domestic drone use.

⁹The specific debate surround the FAA Reauthorization Act are beyond the scope of this essay. For a detailed discussion of domestic drone use see ACLU 2011.

These policies have resulted in substantial profits for the UAV industry. While drones comprise only a part of a larger portfolio of items produced by these firms, UAV contracts have grown immensely since 9/11. Between 2001 and 2011, more than 2,000 contracts, specifically designated for the research, development, or manufacture of drones have been awarded to 21 separate companies by the DOD and other government agencies (USA Spending 2013). In 2001, approximately \$116 million in contracts were awarded for drone production. By 2011, that figure had increased to some \$1.8 billion (*Ibid*). Figure 3 provides a breakdown of the dollar amount of known UAV contracts issued by the DOD between 2001 and 2011.



i gare et brone, erry contracts (in minions of donars) by real

¹⁰ Source: USA Spending 2013

¹¹ These numbers are likely understated. Contracts included in this figure include those which *specifically* reference "drones" or "unmanned aerial vehicles." Other contracts for complementary factors, like robotics systems, are not reported in the above figure. While these are necessary inputs into the manufacturing of UAVs, the reporting system for such

Though already substantial, profits related to the development and manufacture of UAVs industry are anticipated to increase to over \$89 billion per year (Teal Group 2012). One report from the Government Accounting Office (GAO) found that the DOD's acquisition plan for UAVs alone would require some \$34.9 billion, not including the potential costs related to UAV repairs or modifications (Gertler 2012: 33). The potential revenue from additional contracts provides further incentives for those in the defense industry to engage in intense rent-seeking via the aforementioned mechanisms of lobbying and campaign contributions. By continuing to invest in relationships with Congress and advocating certain policies, UAV manufacturers are likely to observe higher profits.

Using this knowledge one may examine more specific cases where the private incentives of these special interests groups worked to create policies which fail to fulfill the public interest. Consider the case of the Block 30 Global Hawk. In response to an announced \$487 billion reduction in the DOD budget, the Air Force proposed to remove the Global Hawks from their fleet to save \$2.5 billion over five years (Sia and Cohen 2013).

In terms of the public interest basic cost-benefit, this decision seems to have been sound. The technical efficiency of the UAVs was questionable. Each unit cost more than \$200 million to produce, nearly 300 percent more than original estimates (Roston 2013).

contracts make it impossible to differentiate between funds used for robotics in UAVs versus other uses.

In discussing the cost of the Global Hawk compared to its main alternative, Air Force Sec. Michael Donley stated, "it's cheaper for us to continue with the [alternatives]" (quoted in Beidel 2012). In addition, the UAV was operationally suspect. In discussing the Global Hawk, a top Pentagon official bluntly stated, "the Block 30 is not operationally effective" (Sia and Cohen 2013). The UAVs encountered numerous mechanical issues and their performance consistently failed to meet efficacy guidelines (Gilmore 2011).

However, the Air Force's plan to cut the Global Hawks was never implemented. Upon learning of the plan to cut the Global Hawk from the Air Force arsenal, Northrop Grumman, the manufacturer of the UAV, launched an intensive campaign to keep the technology in use. With the Global Hawk projected to bring more than \$1 billion in revenue to Northrop Grumman, the firm faced strong incentives to keep the program despite reports the UAV did not fulfill the goals of the Air Force (Northrop Grumman 2012a: 33-34). The firm created a "Support the Global Hawk" website, urging citizens to petition their Congressional representatives (Northrop Grumman 2012b). The firm poured thousands of dollars of campaign contributions and lobbying. Following these efforts, the plan to ground the drone was blocked by Congressional directive. In late 2013, the firm announced it received a \$144 million dollar contract for the three Global Hawks the military stated they no longer wished to purchase (Hennigan 2013).

4.2 Congress

Elected officials, like private contractors, face incentives to expand the use of UAVs in a manner which may not completely align with the public interest. While

providing laws and regulation, politicians seek a combination of wealth, votes, fame, and goods and services for their constituencies. In exchange for these goods, representatives can create laws and control the budget for a variety of bureaucratic organizations. The aforementioned campaign and lobbying contributions from UAV manufacturers are two of the strongest incentives facing lawmakers with regard to drone policy.

The power of these influences is pointedly illustrated by the formation of the Congressional Unmanned Systems Caucus (CUSC) in 2009.¹² The 50 member caucus seeks to,

[E] ducate members of the Congress and public on the...value of [UAVs]; [and] actively support further development and acquisition of more systems....We work with the military...and other[s]...[to] support policies and budgets that promote a larger, more robust national security unmanned system capability (CUSC 2013, emphasis added). Members of the caucus face incentives to expand the use of UAVs, even if such

extensions do not completely align with the broader public interest. First, every member of the caucus comes from a state with a connection to drone manufacturers. Each state represented in the caucus houses at least two of the largest campaign contributing UAV manufacturers. Forty-two percent of caucus members come from states in which all five companies maintain operations (Boeing 2013, General Atomics 2013, General Dynamics 2013, Lockheed Martin 2013, Northrop Grumman 2013). More than half of caucus members represent states projected to reap the largest gains from UAV use (AUVSI 2013: 3). These observable gains are important in this context, as they are one way in

¹² A similar caucus was created in the Senate in 2012 with a mission to "educate Senators and staff on the capabilities and concerns of UAS and to work closely together to best shape the UAS policymaking process (Harder and Heisten 2012). Like the House Caucus, the Senate group is also comprised of members with a particular interest in advancing the industry.

which politicians can fulfill their private goals. New employment and other opportunities demonstrate to constituents that an elected official is bringing benefits to their district and should be reelected. It follows that caucus members have a vested interest in expanding the industry, in order to generate benefits for their constituents and future campaigns.

The potential benefits to Congressional officials for expanding the use of UAVs may be illustrated by several specific examples. Congressman Buck McKeon, one of the caucus' co-chairs, and representative of the district in which Northrop Grumman manufacturers UAVs, for example received over \$375,000 in contributions from the industry between 2008 and 2012 (CRP 2013). His home state of California obtained more than \$4.3 billion in drone-related contracts between 2001 and 2011 and is projected to benefit more than any other state from increased UAV use (AUVSI 2013: 3, USA Spending 2013). Three other caucus members, Ken Calvert (CA), Joe Courtney (CT), and Robert Wittman (VA) each received more than \$100,000 between 2008 and 2012. Other member of the caucus likewise received substantial contributions (CRP 2013).

These benefits for elected officials provides a clear window as to why they may pursue policies which do not align with the public interest. Stated simply, the incentives faced by lawmakers may not align with the broader public interest of the U.S. citizenry as a whole. Congressman McKeon illustrates this idea when discussing cuts to defense spending. Though cutting the federal defense budget may fulfill the desires of the general public, he maintained focus on his specific district.

[T]hese cuts...[threaten my district's] economy....Drastic cuts will hurt many communities in our district...which account for over \$1.4 billion in defense contracts...[I]f these cuts are to take place, all of those contracts could be cut across the board by 8-12%. These institutions alone employ 25,000 people (McKeon 2013). Caucus members and other elected officials have several means to influence drone policy. First, Congress controls a variety of discretionary budgets. Between 2001 and 2003, for example, Congress provided an additional \$38 billion for costs related to the War on Terror (GAO 2003: 1). A significant proportion of these Defense Emergency Response Funds (DERFs) went to UAS projects. General Atomics, for instance received millions in DERF to advance two of its drone platforms. Congress granted some \$235 million to further develop the firm's "Predator" platform (Bone and Bolkcom 2003).

In addition to DERF, the budget for the DOD is controlled by Congress through the annual National Defense Authorization Act (NDAA). The Congressional orders of the NDAA may all but force the use of some systems, even if these systems are undesirable and do not align with the interests of the public. The NDAA of 2007, for example, radically changed the process of UAV acquisition. Prior to this legislation, unmanned aerial systems were designed after a comparable manned system already existed. Following the NDAA for 2007, however, unmanned systems would be the assumed starting point for new technologies (NDAA 2007, section 941). This implies that, except under special circumstances, any new aerial systems are to be designed as unmanned aircraft, despite the reports from the DOD that UAV technology is operationally ineffective in many instances, and thus would not provide the best possible defense for U.S. citizens (see Reed 2013, Zenko 2012). Other NDAA directives likewise appear to promote drone technologies which may not provide the best defense. The NDAA for 2010, for example, required funds to be used to produce the "Sky Warrior" drone despite the UAVs high failure rates and "poor reliability" (Beckhusen 2012, NDAA 2010 §214).

One may observe still other examples of how the incentives facing lawmakers may yield policies counter to the public interest. As noted in section 4.1, the decision by the Air Force to cut the Global Hawks appeared to have looked to fulfill the public interest as resources could have been used to provide superior defense. In discussing the retirement of the Global Hawk, one Air Force officer stated, "Why...mak[e] us [the Air Force] spend money on something we don't want or need?" (quoted in Roston 2013). Air Force Chief of Staff, General Mark Welsh, stated the piloted U-2 was better suited to meet the needs of the Air Force, "We want the platform that will do the best job of accomplishing the mission assigned" (*Ibid*).

While the aforementioned activities by Northrop Grumman were integral in keeping the Global Hawk program, the incentives and actions of Congress were no less important. Upon learning of the plan to cut the Global Hawk, Congressman Buck McKeon, co-chair of the Unmanned Systems Caucus and head of the House Armed Services Committee, promptly sought to block the proposed cuts, arguing the UAV should be kept operational (Sia and Cohen 2013). McKeon, who had received an additional \$10,000 from Northrop Grumman soon after the Air Force's announcement about the Block 30, had immense incentive to keep the project, even though the technology did not look to fulfill the public interest. McKeon represents the district in which Northop manufactures the Global Hawks, meaning his constituency would be directly impacted by any changes to the Global Hawk program. Working with other members of Congress, the NDAA for 2013 *required* the Air Force to continue to maintain the Global Hawks already in use and to purchase those UAVs which were

intended for production. Further, the act prohibited the use of any federal funds to retire or dismantle the drones. (NDAA 2013 §154).

5 Conclusion

This work has four main implications. First, this chapter implies that we cannot be confident that drone technology is being developed and utilized in a way which fully fulfills the public interest. The current institutional structure in which policymakers and special interest interact does not clearly align the interests of the public with the interests of policymakers. This suggests that policies regarding UAVs are subject to the private influences of a variety of actors. This work demonstrates how, even when faced with evidence of technical and allocative inefficiency and the recommendations of experts to alter policy, those involved in constructing drone policy may undertake actions which contradict public interests.

Second, this analysis has implications for drone policy. Given the incentives faced by the various private and public actors for expanding the drone industry, policy regarding drones may be subject to a type of inertia. That is, drones may continue to be developed and manufactured even when conditions (i.e. budgetary changes, differences in type of war, enemy combatant, technological changes, etc.) suggest another technology or method is preferable. Although drones may be desirable in some cases, this is not always the case. Recent reports, for example, have discussed the ineffectiveness of UAVs in situations where counter-aerial technology is available. Shooting down two UAVs frequently used in the War on Terror, has been referred to as "child's play" (Palmer 2012). Despite the fact drones may not be the best means to serve

the public interest in these and other cases, the internal and external pressures faces by policymakers indicate further development and manufacture of UAVs.

Third, this chapter has larger implications for general defense policy. Although drones have become a key component in military operations, they are but one technology in the military arsenal. This work demonstrates how one just one component of national defense may have immense impact on the incentives faced by policy makers and how these incentives may cut against the broader public interest. Given that the rules that govern drone policy are very similar to the rules which govern other defense acquisitions, this indicates that the misalignment of defense policy with the public interest may be substantial.

Fourth and finally, this work calls into question a typical assumption made within the larger literature on defense and peace economics regarding defense provision and security policy. This chapter indicates that the standard narrative of benevolent public actors looking to maximize a larger social welfare function may not be the appropriate lens for analyzing defense issues. Instead, when discussing the research, development, and ultimate use of drones or any other defense technology, this work suggests it is necessary to understand the individual incentives facing both private and public actors.

DRONES COME HOME: FOREIGN INTERVENTION AND THE USE OF DRONES IN THE U.S.

"Government, even in its best state, is but a necessary evil; in its worst state, an intolerable one." –Thomas Paine (1775-6: 69)

"If men were angels, no government would be necessary. If angels were to govern men, neither external nor internal controls on government would be necessary. In framing a government which is to be administered by men over men, the great difficulty lies in this: you must first enable the government to control the governed; and in the next place oblige it to control itself. A dependence on the people is, no doubt, the primary control on the government; but experience has taught mankind the necessity of auxiliary precautions." –James Madison ([1788] n.d.: 337)

1 Introduction

In order for a government to function, individuals must transfer some personal control to the state. The danger in granting the government these powers, however, is that the state may then engage in predatory behavior against the citizenry. The problem with government control, as Paine and Madison's words emphasize, is that a government which is empowered to conduct certain activities, must simultaneously be trusted not to abuse this power. This "paradox of government," the issue of how a government can be simultaneously empowered and yet constrained, is at the heart of constitutional political economy (see Buchanan 1975; Buchanan and Brennan 1985; Weingast 1995; Gordon 2002). While the use of force by governments can, in theory, protect citizens from threats to their person and property, force can also be used by those in power to undermine the very rights they are charged with protecting.

In the days since Paine and Madison's insights, the world has observed astounding technological progress. These technologies have the benefit of advancing civilizations in a variety of capacities. While society has profited greatly from these advancements in terms of wealth creation, these technologies have also lowered the cost of government performing its core activities, as well as expanding what constituted the scope of those activities. Technological advances allow governments to better communicate and coordinate their activities over greater geographic areas (see Cowen 2009). In principle, the existence of strong constitutional constraints, which limit abuses of government power, would ensure that technological advances would not lead to expansions in the scope and scale of government. In practice, however, this may not be the case if technological advances change how government carries out its domestic activities. The purpose of this chapter is to analyze how such a process might work by focusing on the evolution and domestic use of unmanned aerial vehicles (UAVs) or "drones" in the U.S.

We extend and apply the analysis by Coyne and Hall (2014) who consider how coercive foreign interventions can result in an expansion of domestic government powers.¹³ They demonstrate how, even domestically constrained governments, face weak or altogether absent limitations when acting abroad. This allows governments to act in the unconstrained manner that Paine and Madison feared. Coyne and Hall (2014) develop the theory of the "boomerang effect" of foreign intervention, explaining how innovations in

¹³ "Coercive foreign interventions" are defined by the intervening government attempting to shape outcomes—political, economic, social, legal—to achieve an end that is different from what would have emerged absent the intervention. In order to achieve this end the intervening government invests resources to control and suppress any resistance from the foreign population.

state-produced social control can be imported back to the intervening country. When this importation occurs, the result is an expansion in the scope of domestic government activities and loss of liberty for citizens. The logic of the boomerang effect helps explain how drones, originally developed for use in foreign interventions to control distant populations, have slowly been incorporated into domestic government activities.

Drones—aircraft without a human pilot controlled either by remote control or autonomously by computer—have been used for reconnaissance missions abroad since the Cold War (see Hall and Coyne 2014).¹⁴ While first used for reconnaissance and later offensive missions abroad, the use of drones domestically by the government presents a substantial threat to civil liberties in the U.S., as the technology effectively decreases the cost of government-produced social control over U.S. citizens—e.g., surveillance, monitoring and tracking, etc. The current use of drones in the U.S., and the push to expand their domestic deployments further, highlights the potential for government agencies to abuse the technology at the expense of U.S. citizens.

The first use of drones domestically occurred during 2002 by the Customs and Border Protection (CBP). According to Jennifer Lynch, an attorney with the Electronic Frontier Foundation (EFF), not only are the advanced lenses and thermal imaging utilized by drones already in use, but the use of UAVs by government agencies is already raising seriously questions regarding civil liberties and legal issues (see Coia 2013). The activity

¹⁴This definition is consistent with the definitions utilized by the Department of Defense. Under the DoD definitions, a UAV is defined as a single air vehicle while a UAS consists of three to six vehicles, ground control, and support equipment. The term "drone" has become a popular term referring to both UAV and UAS, though it is not utilized by the DoD or private industry.

of drones in the U.S., however, has not been restricted to the areas along the U.S. borders. Between 2010 and 2012, for example, the CBP conducted 687 missions *for other agencies* (Fulton 2014). The CBP has reported that the use of their drones has led to more than 6,800 arrests. According to the FAA, 273 agencies had permission to fly drones as of December 1, 2011.

While the ultimate extent of drone use has yet to be realized in the U.S., the use of the technology by the CBP and other agencies has the potential to undermine domestic liberties. How is it that drones, technology once exclusively used in surveillance and offensive missions abroad, have come to be used on U.S. soil? How is it that, despite concerns regarding safety and privacy, the use of UAV technology domestically has taken root and continues to expand? To answer these questions, we turn to the logic of the "boomerang effect" developed by Coyne and Hall (2014). By applying their framework to the case of drones, we trace out the origins of drone use in the U.S. and identify the mechanism through which UAV technology was imported back into the U.S.

The contribution of this chapter can be situated at the intersection of two literatures. The first is the literature on constitutional political economy (see Buchanan 1975; Brennan and Buchanan 1985; Weingast 1995; Gordon 2002) which explores the role, design, and enforcement of rules as constraints on government and private behavior. The second is the literature on the dynamics of interventionism (Mises 1929, 1949;

Rothbard 1970; Kirzner 1985, Ikeda 1997, 2005) which focuses on how government interventions lead to a series of perverse unintended consequences.¹⁵

Our contribution to these literatures is to explore how foreign interventions can erode and undermine domestic political institutions and constraints on government. We argue that domestic political institutions are not invariant to interventions abroad and that interventions can have long and variable effects which are undesirable from the standpoint of the citizens living in the intervening country. Austrian economists typically focus on the dynamics of intervention within a market context. Our analysis focuses on the chain of consequences from intervention by one government in another society. Particular focus is placed on the negative unintended consequences on domestic political institutions.

We proceed as follows. Section 2 discusses the boomerang effect framework and identifies the four channels through which coercive foreign interventions may generate domestic effects. Section 3 provides a brief historical overview of the use of drones in foreign interventions. Section 4 applies this framework to the case of UAVs and is divided into three subsections. Section 4.1 examines the general implementation of UAVs domestically within the U.S. Section 4.2 examines the specific case of the Predator Drone, arguably the most well-known UAV utilized on American and foreign and soil.

¹⁵ The dynamics of intervention framework was developed to study government intervention in the specific context of the market economy. Given this, our analysis deviates from the pure dynamics of intervention framework. Our focus is on how government interventions in other societies can lead to unintended, and often undesirable, changes in domestic government activities and institutions.
Section 4.3 analyzes how the use of drones threatens the privacy and liberties of the U.S. citizens. Section 5 concludes and offers implications.

2 The Boomerang Effect

A defining characteristic of coercive foreign interventions is that governments push to extend their power and influence to distant geographic territories to shape political, social, economic, or other outcomes. These interventions necessarily require exerting some form of control over the foreign population in order to stifle resistance and achieve the desired effect. This has important implications for the way we think about constraints on government, and the aforementioned paradox of government. In particular, the same constraints that effectively limit government behavior against its own citizens typically do not extend outside a nation's geographic boundaries. The U.S. Constitution, for example, does not extend the rights afforded to U.S. citizens to foreigners.¹⁶ This implies that, even when a government is constrained domestically, foreign interventions allow members of a government to circumvent these constraints when dealing with foreign populations. It follows that in the absence of strong international constraints governments, which are constrained in their actions domestically, are able to intervene abroad with few limits on their power. These interventions may not only impose high costs on foreign populations, but ultimately work to change the character of governmentproduced social control at home. There are four related channels through which changes

¹⁶ Although we focus on the U.S. use of drones in this chapter, the framework of the boomerang effect is not limited to the case of the U.S. The boomerang effect maintains that expansions in the scope of government may occur as a result of engaging in foreign intervention as described above. For additional examples of the boomerang effect and a more detailed discussion of the speed and magnitude of the boomerang effect, see Coyne and Hall, 2014.

state-produced social control abroad may boomerang back into the intervening country (Coyne and Hall 2014).

First, the initial move by government to engage in interventions abroad necessarily changes the composition of domestic government activities. By mobilizing forces to engage in coercive foreign interventions, the government centralizes its decision-making power and activities for the purposes of designing, implementing, and overseeing the intervention (see Jacoby 1973; Higgs 1987, 2004, 2007, 2012; Porter 1994). This initial centralization threatens the checks and balances placed on government as those agencies which were once independent of the political center, now become intertwined in its activities. For example, in the wake of the "War on Terror" many state and local agencies became linked with, and in some cases subsumed by, federal agencies. In the process, these state and local agencies became dependent on the federal government for funding and for directives. The result was that the power of the federal government grew while the separation of activities between different levels of government were weakened, if not altogether removed.

The second channel through which interventions may return to impact domestic government activities is through shaping the human capital of those involved in coercive foreign interventions. Human capital refers to the knowledge and abilities of an individual that contributes to their productivity. Simon (1997) and Merton (1940) posit that participating within an organization and working in a bureaucratic structure influences behavior and ultimately shapes an individual's skills and worldview.

Within the context of foreign interventions, this shaping of human capital has strong implications. In order for interventions to be successful, those undertaking the interventions must either possess or develop the necessary human capital or skills to achieve the desired ends of the intervention. Whether an individual already possesses these skills or acquires them, participation in the intervention provides an incentive to further develop and hone the skills required for foreign intervention. These skills range from a willingness to implement the dictates of the intervening government to more oppressive actions like surveillance, censorship, and violence. In some cases it is existing members of the intervening government that possess or develop the relevant human capital. In other cases the intervening government "imports" foreigners with an existing set of skills in social control from previous foreign interventions in which they participated. In each case the result is the same. Those involved in foreign interventions develop and refine their human capital as specialists in state-produced social control. Following the intervention some of these specialists return to domestic activities and bring with them their refined human capital. With a comparative advantage in certain types of state-produced social control, these specialists will look for opportunities to utilize their unique skills domestically.

The third channel through which the boomerang effect may occur is through changes in administrative dynamics within the intervening country. As per the second channel, those involved in implementing interventions abroad—civilian or military—eventually reallocate their skills to other tasks in the private and public sectors. These skills may be implicit—meaning that the individuals shape the organizational context in

which they operate through employing the skills they acquired abroad. In this case, activities that were previously unacceptable (e.g. the use of a particular tactic or operational method) become normalized as the way that foreign interventions were conducted abroad becomes standard procedure domestically. For example, methods used to combat terrorists abroad may also be used to combat potential terrorism at home in the name of national security.

The skills associated with state-produced social control may also be more explicit—meaning that an individual becomes renowned for being an expert in a particular type of social control (e.g. military tactics, surveillance, etc.). They are able to use, and are rewarded for, implementing their methods domestically. These specialists may be involved in the creation of new agency, or involved within an existing state entity. Regardless of whether their skills are used implicitly or explicitly, changes in administrative dynamics lower the cost of using state-produced social control techniques domestically since these activities become normalized into the operations of the agency and become a part of everyday life.

The final channel of the boomerang effect involves innovations in physical capital resulting from state-produced social control. As Cowen (2009) argues, growth in the scale of government may be, at least in part, attributed to advances in technology that lowered the cost of operating a larger government. This argument also has implications for the *scope* of government activities as well as the scale.¹⁷ As technological innovations

¹⁷ Scale refers to the size of government while scope refers to the range of activities that government undertakes. Scale and scope can be positively correlated, but they need not be.

allow for the government to more cheaply coerce foreign populations, it likewise lowers the cost of using similar techniques domestically. Even if technologies are created with the sole purpose of use abroad—e.g., UAV or "drones"—these technologies may end up being used within the intervening country. Just as the human capital channel allowed for individuals to utilize their skills domestically, advances in physical capital allow for these individuals to further enhance their productivity in carrying out state-produced social control over both foreign and domestic populations. For example, the use of individual agents to surveil and monitor citizens may be politically unpalatable and prohibitively costly. However, the advent of enhanced monitoring techniques like wiretapping, cameras, and drones, effectively lower the cost of engaging in these activities at home both in monetary terms and in terms of being easily observable by citizens.

Taken together, the three aforementioned channels—the human capital channel, the physical capital channel, and the administrative dynamics channels—reinforce the initial centralization observed at the outset of the coercive foreign intervention. Coercive foreign interventions, regardless of their original motive, shift enhanced human and physical capital resources in state control toward the political center. Previously decentralized institutions and groups which served as a check on the actions of those in power now become subservient to, and dependent upon, the political center. Further, the movement of individuals with skills in social control into administrative positions alters the mentality of both public and private organizations where the expanded scope of government activities is seen, not as unacceptable or worrisome as they had been before, but instead become standard operating procedure. If citizens are deferential to the activities of these agencies then the result is that those in the intervening country become more tolerant of expanded government activity as it becomes normalized.

Moreover, this continued movement to the political center generates a reinforcement effect through further changes in human capital and administrative dynamics. Recognizing an opportunity for potential personal gains, individuals with human capital conducive to the expanded activities of the political center—even those who were uninvolved in the initial intervention—will be drawn into positions which utilize these skills. This attraction thus reinforces the changes in human capital and administrative dynamics brought about by the initial intervention.

It is important to be clear that while this framework may work explain expansions in the scope of government activities, it does not follow that such expansions occur *solely* as a result of the boomerang effect. Other types of crises, such as domestic economic downturns, may also contribute to changes in the scale and scope of government (see Higgs 1987). It is also important to note that the framework of the boomerang effect does not imply that the changes observed are irreversible. Instead, the framework suggests that reversing changes is often highly costly due to fundamental changes in the structure of government activities. Moreover, the specific channels at work, as well as the speed and magnitude of these effects will vary across interventions. In some instances the boomerang effect may be substantial while in others it may be negligible. The extent of the boomerang effect depends on three key factors.

The first factor involves the particular methods and technologies used in the intervention. Some methods (e.g. mutilation and murder) may not be easily imported

back into an intervening country due to the observable brutality of these forms of social control. The second factor impacting the overall speed and potency of the boomerang effect involves the ideology of the domestic country. In particular, the dominant ideology of both the citizenry and the judiciary regarding state power will impact both the legality and speed at which the boomerang effect may occur. For example, if a particular government activity is unacceptable to a large segment of the voting population, this may prevent, or at least slow, the use of the particular method domestically. The third factor impacting the important of foreign social control domestically involves the nature of the group targeted by the foreign intervention. In general, if targeted groups are on domestic soil, then it is more likely that state-manufactured social control abroad will be imported back domestically. For example, during WWII, the presence of Japanese citizens in the U.S. prompted the use of internment camps because it was easy for the U.S. government to categorize Japanese-Americans as the enemy. Similarly, the British government called all of citizens of Austrian, Germany, and Italian heritage into special tribunals as these groups were assumed more likely to be spies or provide information to enemies of the British government. Thousands were placed in internment camps or deported throughout the war (BBC 2012).

Finally, it is important to note that this framework does not imply that the boomerang effect is irreversible, nor that the changes associated with the boomerang effect will occur immediately. The theory suggests that reversing the boomerang effect would be highly costly due to underlying shifts in the structure of government activities. Further, the framework does not imply a clear mapping from intervention to immediate

reductions in freedom. Although an intervention may initially appear to have no impact of domestic liberties, these effects may be subject to a substantial lag. As we discuss in the following sections, we may trace the use of drones as a means of foreign intervention back to the Vietnam War, implying that the current use of drones has been evolving for some fifty years.

3 Drones as a Tool of Foreign Social Control

In the broadest sense social control involves the use of rules to govern human behavior. As Ellickson (1987) notes, the source of these rules can come from a variety of sources—personal ethics, agreements between individuals, or from a third party. Coercive foreign interventions entail attempts by a third party, or collection of third parties, to shape the outcome in another society. This requires the intervening government(s) to employ certain tools to produce social control over the population they seek to influence. Only by creating, implementing, and enforcing rules can the intervening government achieve its desired ends. While there are a wide variety of historical tools that have been used for state-produced social control we limit our focus to the use of drones given the growing prevalence of this technology.¹⁸

Although drones have come into the national spotlight only recently, the U.S. military has employed drone technology in foreign interventions for decades.¹⁹ The first U.S. drones were not used as a tool of intervention themselves, but were instead used to

¹⁸ A wide array of methods and technologies may be used to engage in foreign social control. These interventions may be "soft" (e.g. looking to influence the outcomes of the political process, offering various types of aid, etc.) or "hard" (e.g. launching military operations abroad, occupying a foreign territory, etc.).

¹⁹ For a more complete history of the evolution of UAVs, see Hall and Coyne 2014.

train those who would be intervening in combat missions abroad. In 1939, the military acquired what became the first target drones. Both the Army and the Navy utilized the technology as a way to train anti-aircraft gunners. More than 15,000 of these UAVs were used as a training tool throughout the war (Newcome 2004: 58).

During the late 1940s and throughout the 1950s and 1960s, UAV technologies were refined. It was during this period that UAVs were seen, not only as a way to train for intervention, but as a more direct means of carrying out foreign invention. The start of the Cold War between the U.S. and the Soviet Union prompted the U.S. government to invent new ways to spy and gather data on real and perceived enemies with the goal of stopping the spread of communism. In 1955, an earlier drone model was modified to include a series of film cameras for the purposes of reconnaissance. The U.S. Army introduced the enhanced drones in 1959 and eventually used over 1,400 of the UAVs abroad (Newcome 2004: 59). Although no open conflict ever occurred between the U.S. and the Soviet Union, the U.S. attempted to intervene in Russia and gather intelligence via UAVs. In an effort to avoid the political turmoil which would result from airmen being captured and killed by the Soviets, the U.S. embarked on a number of surveillance drone programs (Newcome 2004: 71).

Drones use for reconnaissance missions continued throughout the Cold War and also during Vietnam. However, it was the first Gulf War that would prove critical in the development of drones as a major tool involved in foreign interventions. UAVs were used substantially throughout the short conflict. According to a May 1991 report from the Navy, "at least one UAV was airborne at all times during Desert Storm" (quoted in

Frontline 2013). The conflict saw 522 separate drone launches and over 1,600 hours of flying time. It was during this conflict that U.S. first observed the ability of UAVs to intimidate and control the behavior of both the Iraqi military and civilians. Crews working in Desert Storm reported two cases of groups of Iraqis attempting to directly surrender to unmanned aircraft (Shelsby 1991).

Throughout the 1990s, the military undertook operations in Somalia, Bosnia and Herzegovina, Rwanda, Kosovo, and other places throughout the world. Again, UAVs would play a part of the interventions. Unlike previous military engagements, the motivation was not the threat of an enemy to the U.S. government. Rather, the stated goals were more humanitarian in nature with the interventions intended to end conflicts, among other concerns (see Seybolt 2008). The use of traditional full-scale military operations and equipment proved difficult and often impractical both technically and politically. Drones provided a means through which the military could engage in humanitarian intervention while not engaging in a full scale ground invasion and while keeping soldier fatalities in check. By using drones, many missions would no longer require as many soldiers in the field, if they required them at all. The idea was that drone technology could be used as a tool for shaping the outcomes in foreign societies while reducing the risk of injury and fatality to U.S. soldiers. The overall result was that drones lowered the cost of intervening abroad.

UAVs have seen their most expansive use in the Global War on Terror. Targeted killings by UAVs began in 2002 and U.S. forces have conducted drone missions in Pakistan, Yemen, and Somalia (Sifton 2012, Mazzetti 2013). In additional to the use of

drones for surveillance missions, more than 380 confirmed offensive strikes occurred in Pakistan between 2004 and 2013. As many as 71 strikes were conducted in Yemen and nearly ten in Somalia (Bureau of Investigative Journalism 2014).

Given this historical evolution, the central question is how the transition from the use of drones in foreign interventions to their use on domestic soil occurred. How is it that drones, initially used for target practice, then for surveillance, and later for offensive missions abroad, came to be used as a means of government-produced social control domestically?

4 Drones as an Illustration of the Boomerang Effect

Just as the evolution and use of drones abroad involved a variety of actors and events, so too did the importation and use of drones domestically. While no one event or individual is solely responsible for the use of drones in the U.S., one observes the logic of the boomerang effect in a variety of aspects of the use of UAV technology domestically. These instances illustrate how the four aforementioned channels work to bring foreign methods of social control home. Taken together, we observe how these smaller instances of the boomerang effect, though enacting change in their own right, together greatly alter the cost of and benefits of social control mechanisms domestically and, ultimately, threaten the liberties of U.S. citizens.

4.1 Arthur Cebrowski, Force Transformation, and Domestic Drone Use

There is perhaps no better illustration of the boomerang effect in the evolution of UAVs than the career of Vice Admiral Arthur Cebrowski. Cebrowki worked as a naval officer and president of the Naval War College. He also worked as the head of the Office of Force Transformation within the DoD (see Singer 2009, Springer 2013). While the development and implementation of drones for international and domestic uses in the U.S. has a nuanced history (see Hall and Coyne 2014), Cebrowski has been referred to as the man, "directly responsible for the rise in the number of robots and drones within the U.S." (Springer 2013: 145). The changes observed in the use of autonomous technologies within the DoD and elsewhere may be attributed, in significant measure, to Cebrowski's time working abroad for the U.S. military and his efforts to implement his ideas.

Cebrowski graduated from Villanova in 1963 as a member of the Naval Reserve Officer Training Corps (NROTC). Following his graduation, Cebrowski was deployed in two tours of duty in the Vietnam War. He flew 154 missions, most over North Vietnam (Blaker 130). Throughout the 1970s and 1980s, Cebrowski rose through the ranks of Naval Aviation and continued to acquire significant flight skills. It was through these experiences that he became interested in large scale integrated systems-complex computer and information sharing systems-because of their potential use in combat, increased technical reliability, and their potential to decrease pilot risk. He stated that his interest in such technology, and desire to expand the use of such technological systems was a direct result of his combat experience, thus illustrating the human capital channel of the boomerang effect. Through Cebrowski's time in Vietnam, he acquired a variety of skills and ideas regarding autonomous technology and combat. Upon returning to the U.S., he brought these experiences and skills with him, and, as the changes in administrative dynamics channel would suggest, utilized his talents in his subsequent positions. He noted:

My experience had come from flying combat aircraft....It was a complex situation, and success depended on manipulating that complexity....Transistors and the large-scale integrated circuits [complex computer systems]...promised to shift some of that complexity [from the pilot] to the aircraft and the weapons it carried. [If the] military could do that on a large scale...[it] would have an edge. It could lead to a strategic advantage, an ability to shift the terrible burden of warfare complexity and the risks it carried away from you onto your opponent (quoted in Blaker 2006: 132) In 1981, the Navy created the Strategic Studies Group at the Naval War College.

The College selected a set of "war fighters," individuals whose careers had kept them employed in tactical as opposed to administrative operations, to propose possible enhancements to the U.S. Navy. During his year-long appointment, Cebrowski began to form a series of ideas regarding the future of military combat, specifically the need to incorporate more enhanced technology into aviation. It was during this period that he began to see change in military affairs as essential (Blaker 2006: 133-34).

The following year, Cebrowski returned to fleet operations. In 1990, he commanded the USS Midway during Operation Imminent Thunder in the First Gulf War. Cebrowski's experiences during Imminent Thunder and Operation Desert Storm are illustrative of how the tactics he encountered and employed would later be brought home. Over a period of six days during Imminent Thunder, the U.S. conducted a variety of drills and operations utilizing an array of techniques and new technology. In an effort to deceive and intimidate Iraqi forces into moving their efforts to the coast of Kuwait, the U.S. conducted numerous military drills and launched an intense media campaign to publicize the operations.

Cebrowski would continue to command the Midway through Operation Desert Storm. During this time, the Navy would launch 228 sorties and more than 100 Tomahawk missiles in order to drive Iraqi forces from Kuwait. It was during Desert

Storm that the military observed the potential of drone technology, not only to conduct surveillance, but to intimidate the enemy. During Desert Storm, the crew of the USS Missouri, one of the ships which had worked in conjunction with the USS Midway during Imminent Thunder, had launched surveillance drones over Iraqi territory. The crews would later report two cases of Iraqis attempting to surrender to the unmanned aircraft (Shelsby 1991). This illustrated to those in the military the ability of drone technology to serve as tool of direct social control.

This assignment and those that followed worked to solidify Cebrowski's commitment to bringing enhanced flight and other technology to U.S. forces. Discussing one of his flight assignments during the period, he stated:

Those assignments...[were] an epiphany or sorts. It was the first time aircraft I piloted that really flew itself...the information that was available in the cockpit was so much better...[it was a movement] from the physical to the information and the cognitive realms. The limit on what the pilot could do was no longer a matter of physical strength or reflexes. The real limit was the level of awareness and knowledge the pilot had....That awareness turned on information...It was no longer the airplane or its pilot that counted the most; it was...[the] networked environment (quoted in Blaker 2006: 134). Cebrowski spent the next decade following the First Gulf War attempting to bring

this idea of technological advancement to U.S. forces. By the mid-1990s, then serving as Director of Command Control, Communications, and Computers on the Joint Staff, he pushed for the expansion military technology, arguing that superior information, and not the mass of the military, was essential for success. Given changes in warfare, simply maintaining large numbers of personnel was no longer practical or effective compared to enhanced situational technologies (Blaker 2006: 136).

Through this appointment as the Director of Command Control, and the appointments that followed, one observes the intersection of the human capital,

administrative dynamics, and physical capital channels discussed in Section 3. Cebrowski reallocated the skills he acquired while engaging in the tactical operations for the Navy to administrative positions within organizations like the Naval War College and Department of Defense. These administrative positions, in conjunction with the skills Cebrowski acquired during foreign interventions, would prove critical in the implementation of enhanced drone technology abroad and at home.

Drawing from his experiences in Vietnam and the Persian Gulf, Cebrowski developed and proposed the idea of "network-centric warfare," the thought that conflict favors those groups which are able to acquire and utilize a large information network rather than brute force. This network provides nearly real-time shared imagery of a military situation, disperses this information widely, and uses it as the basis for operations. Such technology, he argued, could be used by a variety of defense and security agencies to engage in both international and domestic projects (see Freidman 2009).

Following an appointment as the President of the Naval War College, Cebrowski was selected by Defense Secretary Donald Rumsfeld to become the director of the Office of Force Transformation (OFT). It was during this period that the federal government began to conduct operations as part of the War on Terror. This "war" was unique in that it was fought on a truly global basis and included not just interventions in other countries, but also increased government interventions on domestic soil with the goal of combatting potential terrorism. Further, the War on Terror had a centralizing effect with the federal government taking the lead, both domestically and internationally, to combat terrorist

threats. The onset of the War on Terror provided the opening necessary for the expansion of drone use both internationally and domestically.

As head of the OFT, Cebrowski was to serve as the "conceptual engine" of the Office, and was tasked to assess and boost "tranformational" programs (Blaker 2006: 138). The OFT was a new agency within the DoD. The newness of the organization combined with Rumsfeld's confidence in Cebrowski and ample resources dedicated to transforming U.S. military capabilities meant that Cebrowski had significant discretion in establishing the size, location, budget, and operations of the office. Under Cebrowski the agency was dedicated to transforming military operations in the 21st century. Illustrating the centralization brought about by the start of the War on Terror, one expressed goal of the OFT was to find new ways to cooperate and coordinate with other federal, state, and local agencies so as to respond to potential threats. In addition to this goal, the OFT actively pursued changes in the physical capital used in combat. As part of this goal, the OFT was integral in the development and use of automated systems, even deploying personnel from technology companies abroad with troops to further develop and refine their products (see DoD 2002: 3, Mobbs 2014).

As a result of Cebrowski's efforts and work to implement enhanced autonomous and information technologies, the use of drones expanded immensely. It was during his time at the OFT that the use of enhanced information systems and UAVs began its greatest expansion. The first targeted killing abroad with a UAV occurred in 2002 and the U.S. Border Patrol used drones for the first time soon after (Customs and Border Patrol Today 2004). According to biographer James Blaker, Cebrowski was "surprised at the

influence his views enjoyed" during his time at the OFT (2006: 139). Specifically,

Cebrowski noted that:

I was surprised by how important the press and other media were to...transformation and how much they became allies in the effort....[We used] the press to make our arguments....[Y]ou get ideas into the audiences who will ultimately determine how far the ideas will go....I worked for the most part ahead of policy, pointing out possibilities. That is appealing to the press and defense contractors....It was a synergistic relationship that turned out to be surprisingly effective (Ibid: 139-140).

This shift in mentality is yet another example of the boomerang effect at work

within the case of UAVs. Drones and other autonomous technologies, once unfathomable, had become a regular part of operations. Their establishment and use internationally had been integrated into the mainstream, recognized by Cebrowski, the OFT, military, and the broader public. This mentality shift created an opportunity for drones to be used not just in an international context, but domestically as well. Drones have been used for surveillance along the U.S. borders since 2002. Not only was the technology intended to prevent and deter the crossing of illegal immigrants, but to stop the movement of individuals with potential connections to countries with known terrorist ties (see Perry 2008, Operation Border Star 2014).

Cebrowski retired from his position as head of the Office of Force Transformation in 2005 and died shortly after. Though not solely responsible for the increased use of drones in U.S., it is clear that Cebrowski played a critical role in drones and their underlying support systems. Cebrowski's time engaged in coercive foreign interventions in Vietnam and the Persian Gulf were essential to the formation of his personal human capital and his later roles in various administrative capacities.

4.2 The Rise and Return of the Predator

The General Atomics MQ-1 Predator has been used extensively throughout the Global War on Terror by the United States and "has become the Department of Defense's most recognizable UAV" (Gertler 2012: 33). The Central Intelligence Agency (CIA) has a known stock Predator drones while the DoD employs nearly 200 (*Ibid*: 8). The extensive use of the Predator has facilitated the development of closely related and well-known UAVs, including the Grey Eagle (also known as the Sky Warrior) and Reaper (Predator B) (*Ibid*: 34-36). In addition to its use abroad, the Predator has been one of the most extensively used drones within the domestic U.S. The Department of Homeland Security (DHS), for example, has modified ten Predator drones for surveillance along the nation's borders and hopes to have acquired 24 Predators by 2016 (Sengupta 2013, ACLU 2011: 6). The movement of the Predator onto U.S. soil is yet another illustration of the boomerang effect.

In order to discuss the development of the Predator, we must first turn to an intervention conducted, not by U.S., but by Israeli forces during the Yom Kippur War in 1973. Looking for a way to draw anti-aircraft fire away from their pilots, the Israeli Air Force sought to develop the physical capabilities of drones. If the military could draw fire toward an unmanned craft, they would be able to more effectively reduce pilot fatalities and damage to their manned aircraft. At the center of the development of these aircraft was Abraham Karem. Known as "the dronefather," Karem was trained as an aeronautical engineer and worked in the Israeli Air Force as an engineering officer (the Economist 2012, National Academy of Engineering 2014). During the conflict, Karem worked on a several projects, leading the development and deployment of a variety of reconnaissance

and other aerial systems, including the target UAV. It is here we see the first examples of the boomerang effect in the case of the Predator. In particular, this experience allowed Karem to develop not only the physical capabilities of UAVs, but further hone his own human capital and establish a reputation for his work on UAVs. Karem's team was not only able to produce a decoy drone, but accomplished the task in less than a month (*Ibid*).

Following his employment with the Israeli Air Force, Karem joined Israel Aircraft Industries (IAI), a government-owned aerospace and aviation manufacturer that produces aerial systems for both military and civilian uses (IAI 2014). Although he was poised to be named an Executive Vice President of the company, Karem departed IAI and immigrated to the U.S. where he, "knew opportunities for entrepreneurs were greater" (Whittle 2013). Illustrating how specialists in social control can be imported, Karem took a position with Developmental Sciences, Inc., a U.S. firm who had offered the Israeli government a drone decoy in the 1970s and was working on projects to develop new UAV technologies for government use (*ibid*).

To understand the demand for the type of human capital possessed by Karem, one must appreciate that the War on Drugs undertaken by the U.S. government meant that numerous agencies were now deeply entangled with drug interdiction policy.²⁰ This shift created opportunities for individuals with highly-valued skills (e.g. creating technologies which could be potentially used for surveillance of drug cartels and other groups) to reap

²⁰ A complete discussion of the changes in administrative dynamics brought about by the War on Drugs is beyond the scope of this essay. For a discussion of changes in government agencies as a result of the War on Drugs, see Boettke, Coyne, and Hall (2013), Drug Enforcement Administration (2013), Hall and Coyne (2013), Reuter, Crawford and Cave (1998).

substantial personal gains. It was within this context in the U.S. that Karem was able to utilize, and further develop, his human capital and the physical capabilities of UAVs while assisting the U.S. in engaging in a variety of interventions, including the international war on drugs and multiple humanitarian missions.

With funding from the Defense Advanced Research Projects Agency (DARPA), Karem made additional advancements in the technological abilities of drones and subsequently produced the Albatross, Amber, and Gnat, the precursors of the Predator. Championed by members of the U.S. Southern Command, the drones were intended to provide surveillance of the activities of South American drug cartels. Several years later, the UAVs were in further demand as the CIA looked to closely monitor the conflict situation in the Balkans. Karem met the CIA's demand by modifying the previous drones, thus creating the first Predator (*Ibid*).

Although Karem would leave General Atomics, the current producer of the Predator, before the UAV saw mass production, he played an integral part in its development. Furthermore, the production of the UAV by General Atomics after his departure is illustrative of how the human capital channel can operate. Not only had Karem's previous endeavors on UAVs worked to hone his own human capital, but Karem trained a variety of younger, talented engineers. These engineers then went on to influence the direction of research and development in other firms. For example the first team of engineers assembled by Karem in the U.S., are presently executives and technical experts at General Atomics, including President Frank Pace who Karem once described as, "the closest thing to my right hand man" (quoted in Whittle 2013).

The extensive use of the Predator began with the Global War on Terror and Operation Enduring Freedom and Operation Iraqi Freedom. In February 2001, Hellfire missiles were test-fired from the Predator. Following September 11, 2001, and the subsequent operations by the U.S. military in the Middle East, the U.S. military began to use the Predators not only for surveillance and reconnaissance, but to actively target and kill individuals in Afghanistan and Pakistan. Approximately 115 targets were hit by Predator strikes in their first year of combat operations. The CIA began using armed Predators to conduct strikes in 2002, and a Predator is credited with killing an al Qaeda operative in Yemen. Armed Predators began patrols in Iraq and destroyed Iraqi defenses prior to the U.S. invasion (Callam 2010). The Predator and two of its descendants, the Sky Warrior and the Reaper are the three armed UAVs still in use in the Global War on Terror today. They continue to see use in Pakistan, Yemen, and Somalia and saw extensive use in Libya in 2011.

The domestic use of drones began during the same period. Just as the armed Predators were eliminating international targets, unarmed versions were flying in the skies above the U.S. By 2005, the Customs and Border Protection (CBP) began launching Predators along the borders of the U.S. Although the Predators and other drones have been touted as beneficial for other missions, the CBP demanded the technology as a means to combat the War on Drugs and War on Terror on American soil. Just as there was demand for Karem's drones for use in the international War on Drugs and War on Terror, the modern Predator is now being used domestically for similar purposes. In other words the use of drones internationally has boomeranged back to the U.S. and drones are

not being utilized domestically to combat the drug trade and potential terrorist threats. Members of Congress have stated the domestic use of drones is "invaluable" to operations along the U.S. border (see Booth 2011). Arizona Governor Jan Brewer even declared the CBP Predator drones, "ideal for border security and counter-drug missions" (quoted in Booth 2011). Finally, Texas Governor Rick Perry called for a greater presence of Predators along the U.S. border as part of the state's efforts to combat illegal immigration, terrorism, and the drug trade (see Booth 2011).

We see that the Predator is a particularly pointed example of the boomerang effect. The changes in human capital and administrative dynamics illustrated by the work of Arthur Cebrowski and the OFT, as well as the changes in physical capital resulting from the construction and ultimate use of Predator drone illustrates how UAVs have evolved. Though once developed and used exclusively for missions abroad like the war on drugs and war on terror, we observe how the channels of the boomerang effect have allowed for the expansion of UAV technology domestically. However, the Predator is not the only UAV which has come to see use within the borders of the U.S. In the following section, we analyze how not only Predators, but a variety of UAVs, have been used domestically and, how the use of such technologies threatens the privacy and other liberties of U.S. citizens.

4.3 Implications of the Boomerang Effect: Domestic Drones and the Threat to Privacy and Liberty

As illustrated above, the use of drones in a variety of foreign interventions abroad have been imported back to the U.S. The importation of this technology domestically is not prima facie negative. However, an examination of the domestic uses of UAVs has important implications for the civil liberties of U.S. citizens. Drone technologies have lowered the cost of government monitoring of U.S. citizens domestically and abroad. No longer does surveillance require personnel to directly monitor a target. Targets may now be tracked from a distance, with technology that far surpasses prior capabilities. Highpowered lenses and more advanced cameras, for example, mean that UAVs are better able to observe and monitor an individual's movements from farther away. Night vision, also known as thermal imaging, and Synthetic Adaptive Radar, allows drone operators to see through dense foliage and clouds, as well as walls (ACLU 2011:5). A report on domestic drone use by the American Civil Liberties Union (ACLU), captures the potential impact of these lowered monitoring costs and the use of drone technology domestically on civil liberties.

[M] anned aircraft...has always imposed a natural limit on the government's aerial surveillance capability. Now that surveillance can be carried out by unmanned aircraft, this natural limit is eroding. The prospect of cheap, small, portable flying video surveillance machines threatens to eradicate existing practical limits on aerial monitoring and allow for pervasive surveillance, police fishing expeditions, and abusive use of these tools in a way that could eventually eliminate the privacy Americans have traditionally enjoyed in their movements and activities (ACLU 2011: 1).

These concerns are not without merit. The issues surrounding drones and civil

liberties in the U.S. have already begun to manifest. As noted above, the CBP has used drones substantially along the U.S. borders since the early 2000s and plans to augment their arsenal of UAVs in the coming few years. The use of drones along the border has already raised questions regarding the legality of drones to track individuals. According to Jennifer Lynch, an attorney who specializes in issues of government transparency and privacy, "Customs and Border Protection definitely does use thermal imaging on its drones…I don't think [government agencies] are getting warrants before they are using

thermal imaging. That will be a big issue in the future if they're using it to see what's going on inside a home" (quoted in Coia 2013).

The government use of UAVs, and the subsequent concerns regarding privacy, have not been restricted to the areas along the U.S. borders. In fact, between 2010 and 2012, the CBP conducted 687 missions for other agencies (Fulton 2014). CBP has reported that the use of their drones has led to more than 6,800 arrests. According to the Federal Aviation Administration (FAA), 273 agencies had permission to fly drones as of December 1, 2011, including U.S. Secret Service, Federal Emergency Management Agency (FEMA), Immigration and Customs Enforcement (ICE), and the FBI, for surveillance and other purposes (Department of Homeland Security: Office of the Inspector General 2012: 6). Among these agencies were a variety of state and local police departments. For example, the police department in Miami, Florida requested permission from the FAA to fly drones with their elite special-response team (a version of SWAT). Officials stated the technology would be used to obtain real-time imagery of hostage situations or in standoffs with criminals inside buildings (Dreazen 2011). In June 2011, the Nelson County Sheriff in North Dakota, along with State Highway Patrol, a regional SWAT team, and a bomb squad, utilized a CBP drone with thermal imagining to collect information on three U.S. citizens who were eventually arrested. The CBP used the drone to fly more than two miles above the farm where the men were located in order to determine their exact location and whether or not the men were armed. One of the men involved was later convicted, further adding to the questions regarding the use of drones in police activities (see Coia 2013, Wolverton 2014). In Houston, local television crews

recorded police engaging in secret tests of a UAV. It was ultimately revealed by a police spokesperson that the UAVs could be used in the future to issue traffic citations (ACLU 2011: 7).

In June, 2013, FBI Director Robert Mueller acknowledged before Congress that the agency uses UAVs within the U.S. for surveillance. Though he did not reveal how many UAVs the FBI had at its disposal, or how frequently they had been used, an FBI representative stated that the drones, "[allow] us to learn critical information without introducing serious risk to law enforcement personnel" (quoted in Cratty 2013). When asked whether or not the Bureau had guidelines for using drones which considered individual privacy, Mueller indicated that the agency was in the initial stages of developing their guidelines. When pressed regarding the FBI's policy for obtaining and retaining images gathered from drones, Mueller stated he would have to inquire regarding the FBI's policy (*Ibid*). Since this time, the FBI has admitted to using UAVs at least 10 times since 2006. While the agency wouldn't elaborate on the cases in which drones were used, they stated the drones were "strikingly successful" (quoted in Musgrave 2014).

Perhaps the greatest illustration of the potential harm to civil liberties by drones in the U.S. comes from statements by U.S. Attorney General Eric Holder. In answering questions regarding the use of drones to target U.S. citizens, Holder stated that, "the President could conceivably have no choice but to authorize the military to use such force [targeted killings] if necessary to protect the homeland....[I]t is possible...for the president to authorize the military to use lethal force within the territory of the United States" (quoted in Swaine 2013).

Without clear legal precedent regarding the acquisition and use of UAVs domestically, the outcome for civil liberties is at best unclear. A variety of legal scholars and others have questioned how the use of drones domestically conflicts with citizens' constitutional right against unlawful search and seizure by the fourth amendment. What is clear is that the enhanced surveillance capabilities of UAVs, combined with unclear restrictions on their use, has increased the reach of the U.S. government into the lives of its citizens raising a host of issues regarding the constitutional scope of government activities (see ACLU 2011, Olvito 2013, Thompson 2013, Wolverton 2012).

5 Conclusion

Our analysis demonstrates how drones, although developed for use in interventions abroad, have created a scenario in which domestic individual liberties are threatened. This highlights a broader implication of the boomerang effect—operations conducted outside of a nation's geographic territory may have substantial domestic repercussions. As the case of drones highlights, the innovations in human and physical capital can return home and permanently alter the composition of government activities. Drones have been hailed as a means of protecting the rights and safety of U.S. citizens through their use abroad. What this analysis shows, however, is that even if drones are effective in their international operations, the technology may in fact undermine the very rights they are intended to protect.

This leads us to propose several areas of further research. First, as noted above, the speed and magnitude of the boomerang effect is determined by a variety of factors including domestic ideology, technology, and the nature of the group targeted by the foreign intervention. One avenue of possible inquiry would be an in-depth examination of each of these factors to determine precisely how these issues play into the broader boomerang effect. A second path for future research would be to examine cases of the boomerang effect throughout different periods of time in different countries. Current illustrations of the boomerang effect (see Coyne and Hall, 2014) focus on instances of the framework as they pertain to the U.S. Though illustrative, additional studies of boomerang effect across time and place would work to further develop the theory and illustrate the prevalence of the effect across a variety of interventions.

It is important to reiterate that the presence of the boomerang effect does not immediately imply that the rights and liberties of domestic citizens are doomed. There are several mechanisms which may weaken or prevent the boomerang effect from taking hold. These include: The ideology of citizens, rulings by courts, and evolving legal structures and standards. These factors have the potential to undermine or slow the boomerang effect as it relates to the use of drones domestically.

While the impact of UAVs in the U.S. is still unfolding, it is clear that the origins of drones in the U.S. can be traced to interventions abroad. Through the channels of the boomerang effect, drones have come home. The question that remains is the one which has motivated a variety of scholars both past and present. If, as Paine said, "Government, even in its best state, is but a necessary evil; in its worst state, an intolerable one," the questions remain: How does one keep the state necessary without risking the intolerable? How may we capture the potentially positive aspects of UAVs, while preventing the erosion of liberty?

REFERENCES

- Abizaid, John P. and Rosa Brooks. 2013. "Recommendations and Report on the Task Force on US Drone Policy." *The Stimson Center*. Available online at: http://www. stimson.org/images/upload/researchprfs/task force report FINAL WEB 062414.pdf
- Acemoglu, Daron and James A. Robinson. 2013. "Economics vs. Politics: Pitfalls of Policy Advice," *Journal of Economic Perspectives* 27(2): 173-192.
- Ackerman, Spencer. 2013. "US Drone Strikes More Deadly to Afghan Civilians than Manned Aircraft—Adviser: Study Focusing on One Year of Conflict Contradicts Claims that Robotic Planes are More Precise than Manned Counterparts." *The Guardian*, July 2. Available online at: http://www.theguardian.com/world/2013/jul/02/us-drone-strikes-afghan-civilians.
- Adams, Gordon. 1981. *The Iron Triangle: The Politics of Defense Contracting*. Piscataway, New Jersey: Transaction Publishers.
- Aero News Network. 2012. "Senate Forms UAS Caucus: Bipartisan Group to Help Shape Policy Concerning Unmanned Aerial Vehicles." September 29. Available online at: http://www.aero-news.net/index.cfm?do=main.textpost&id=b7e319f6-0c82-4812-a04e-517a672e2582
- Alker, Valerie. 2013. "Drone Production Amps Up in Florida." WGCU NPR News. January 24. Available online at: http://news.wgcu.org/post/drone-productionamps-florida
- American Civil Liberties Union. 2011. "Protecting Privacy From Aerial Surveillance: Recommendations for Government Use of Drone Aircraft." Available online at: http://www.aclu.org/files/assets/protectingprivacyfromaerialsurveillance.pdf.
- American Civil Liberties Union. 2014. "Status of 2014 Domestic Drone Legislation in the States." Available online at: https://www.aclu.org/blog/technology-andliberty/status-2014-domestic-drone-legislation-states

- Anderton, Charles H. and John R. Carter. 2007. "A Survey of Peace Economics," pp. 1211–1258 in Todd Sandler and Keith Hartley, eds., Handbook of Defense Economics. Vol. 2. Amsterdam: Elsevier.
- Anderton, Charles H. and John R. Carter. 2009. *Principles of Conflict Economics: A Primer for Social Scientists*. New York: Cambridge University Press.
- Arkin, Ronald. 2010. "The Case for Ethical Autonomy in Unmanned Systems." *Journal* of Military Ethics, 9(4): 332-341.
- Arrow, Kenneth J. 1995. "Some General Observations on the Economics of Peace and War, Peace Economics." *Peace Economics, Peace Science, and Public Policy*, 2(2): 1–8.
- Associated Press. 2013a. "Congress Looks to Limit Drone Strikes." February 6. Available online at: http://www.cbsnews.com/news/congress-looks-to-limitdrone-strikes/2/
- Associated Press 2013b. "Lawmakers in Both Parties Urge Oversight of U.S. Drone Program." February 10. Available online at: http://www.foxnews.com/politics/2013/02/10/lawmakers-urge-oversight-usdrone-program/
- Association for Unmanned Vehicle Systems International. 2013. "The Economic Impact of Unmanned Aircraft Systems in the United States." AUVSI. Available online at: http://www.auvsi.org/econreport.
- Association for Unmanned Vehicle Systems International. 2013. "The Economic Impact of Unmanned Aircraft Systems in the United States." AUVSI. Available online at: http://www.auvsi.org/econreport.
- BBC. 2012. "WW2 People's War. Fact File: Civilian Internment, 1939-1945." Available online at: http://www.bbc.co.uk/history/ww2peopleswar/timeline/factfiles/nonflash/a66518 58.shtml
- Beckhusen, Robert. 2012. "'Gray Eagle' Drone Fails All the Time, But Army Still Wants More." *Wired*, June 15. Available online at: https://web.archive.org/web/20130622184812/http://www.wired.com/dangerroom /2012/06/grey-eagle/.
- Beidel, Eric. 2012. "U-2, Global Hawk Advocates Square Off in Budget Battle." *National Defense*. Available online at:

http://www.nationaldefensemagazine.org/archive/2012/May/Pages/U-2,GlobalHawkAdvocatesSquareOffinBudgetBattle.aspx.

- Bennett, Chris. 2013. "Drones Begin Decent on US Agriculture." *Western Farm Press*. Available online at: http://westernfarmpress.com/blog/drones-begin-descent-usagriculture
- Berteau, David, Guy Ben-Ari, Greg Sanders, David Morrow, Jesse Ellman. 2012. "U.S. Department of Defense Services Spending and the Supporting Industrial Base, 2000- 2011." Defense-Industrial Initiatives Group, Center for Strategic and International Studies. Available online at: http://csis.org/files/publication/120524_DIIG_Defense_Service_Contract_Trends. pdf.
- Blaker, James. 2006. "Arthur K. Cebrowski: A Retrospective." Naval War College Review, 59(2): 129-145.
- Boeing. 2013. "Careers: All United States Jobs." Available online at: http://jobsboeing.com/united-states.
- Boettke, Peter J., Christopher J. Coyne, and Abigail R. Hall. 2013. "Keep Off the Grass: The Economics of the Prohibition and U.S. Drug Policy." *Oregon Legal Review*, 17(4): 1069-1095.
- Bone, Elizabeth and Christopher Bolkcom. 2003. "Unmanned Aerial Vehicles: Background and Issues for Congress." Available online at: http://www.fas.org/irp/crs/RL31872.pdf.
- Booth, William. 2011. "More Predator Drones Fly U.S.-Mexico Border." *The Washington Post*. December 21. Available online at: http://www.washingtonpost.com/world/more-predator-drones-fly-us-mexicoborder/2011/12/01/gIQANSZz80 story.html

Boulding, Kenneth E. 1945. The Economics of Peace. New York: Prentice-Hall.

Boulding, Kenneth E. 1978. Stable Peace. Austin, TX: The University of Texas Press.

Bowers, Peter. 1989. Boeing Aircraft Since 1916. Annapolis, MD: Naval Institute Press.

Boyle, Ashley. 2012. "The US and its UAVs: A Cost-Benefit Analysis." American Security Project, July 2. Available online at: https://www.americansecurityproject.org/the-us-and-its-uavs-a-cost-benefitanalysis/.

- Boyle, Michael J. 2013. "The Costs and Consequences of Drone Warfare. *International Affairs* 89(1): 1–29.
- Boyle, Michael J. 2013. "The Costs and Consequences of Drone Warfare. *International Affairs*, 89(1): 1–29.
- Brauer, Jurgen and J. Paul Dunne. 2012. *Peace Economics: A Macroeconomic Primer for Violence-Afflicted States*. Washington, D.C.: United States Institute of Peace Press.
- Brauer, Jurgen and Raul Caruso. 2012. "Economists and Peacebuilding." In Roger MacGintry (ed.), *Handbook on Peacebuilding*. London: Routledge.
- Brennan, Geoffrey and James M. Buchanan. 1985. *The Reason of Rules: Constitutional Political Economy*. Cambridge: Cambridge University Press.
- Brennan, Geoffrey and James M. Buchanan. 1985. *The Reason of Rules: Constitutional Political Economy*. Cambridge: Cambridge University Press.
- Brennan, Geoffrey, and James M. Buchanan. 1985. The Reason of Rules: Constitutional Political Economy. Cambridge: Cambridge University Press.
- Brennan, John. 2012. "The Efficacy and Ethics of U.S. Counterterrorism Strategy." Available online at: http://www.wilsoncenter.org/event/the-efficacy-and-ethicsus-counterterrorism-strategy.
- Brinkerhoff, Noel. 2011. "Drones are a Growth Industry." *AllGov.* October 27. Available online at: http://www.allgov.com/news/us-and-the-world/drones-are-a-growth-industry?news=843487
- Brooks, Rosa. 2012. "Take Two Drones and Call Me in the Morning: The Perils of Our Addiction to Remote-Controlled War." *Foreign Policy*. September 12. Available online at: http://www.foreignpolicy.com/articles/2012/09/12/take_two_drones_and_call _me_in_the_morning.
- Brown, Shannon A. 2005. "Providing the Means of War: Historical Perspectives on Defense Acquisition, 1945-2000." Washington, D.C.: Library of Congress.
- Buchanan, James M. 1975. The Limits of Liberty: Between Anarchy and Leviathan. Chicago: University Press.
- Buchanan, James M. 2003. "Public Choice: Politics Without Romance," *Policy* 19(3): 13-18.

- Buchanan, James M. 2003. "Public Choice: Politics Without Romance," *Policy* 19(3): 13-18.
- Buchanan, James M. and Gordon Tullock. 1962. *The Calculus of Consent: Logical Foundations of Constitutional Democracy*. Ann Arbor: University of Michigan Press.
- Bumiller, Elisabeth. 2011. "Air Force Operators Report High Levels of Stress." The New York Times. December 18. Available online at: http://www.nytimes.com/2011/12/19/world/asia/air-force-drone-operators-showhigh-levels-of-stress.html.
- Bureau of Investigative Journalism. 2014. U.S. Covert Actions in Pakistan, Yemen, and Somalia. Available online at: http://www.thebureauinvestigates.com/2014/04/01/march-2014-update-us-covertactions-in-pakistan-yemen-and-somalia/
- Burris, Keith C. 2013. "Drones Save U.S. Lives." *West Hartford News*, February 25. Available online at: http://www.westhartfordnews.com/articles/2013/02/25/opinion/doc512bd686c82f a867665376.txt?viewmode=3.
- Butos, William and Roger Koppl. 1999. "Hayek and Kirzner at the Keynesian Beauty Contest." *Journal des Economistes et des Etudes Humaines*, 9(2/3): 257–275.
- Byers, Dylan. 2013. "McChrystal on Drones: 'A Covert Fix for a Complex Problem."" *Politico*, February 15. Available online at: http://www.politico.com/blogs/media/2013/02/mcchrystal-on-drones-a-covert-fixfor-a-complex-problem-157126.html.
- Byman, Daniel. 2013. "Why Drones Work: The Case for Washington's Weapon of Choice." *Foreign Affairs*, 29(4): 32-43.
- Callam, Andrew. 2010. "Drone Wars: Armed Unmanned Aerial Vehicles." *International Affairs Review*, 18(3). Available online at: http://www.iar-gwu.org/node/144
- Carey, Bill. 2014. "Cost Overrun Brings Scrutiny to MQ-8 Fire Scout." Aviation International News, May 2. Available online at: http://www.ainonline.com/aviation-news/ain-defense-perspective/2014-05-01/cost-overrun-brings-scrutiny-mq-8-fire-scout.

- Center for Responsive Politics. 2013. Various Pages. Available online at: http://www.opensecrets.org
- Center for Responsive Politics. 2013. *Various Pages*. Available online at: http://www.opensecrets.org/.
- Coia, David Alan. 2013, "US Domestic Drone Use Sidesteps Warrants for Thermal Imaging." *Newsmax*, March 23. Available online at: http://www.newsmax.com/US/drones-warrants-thermalimaging/2013/08/11/id/519767/
- Congressional Unmanned Systems Caucus. 2013. "Mission and Main Goals." Available online at: http://unmannedsystemscaucus.mckeon.house.gov/about/purpose-mission-goals.shtml.
- Congressional Unmanned Systems Caucus. 2013., "Mission and Main Goals." Available online at: http://unmannedsystemscaucus.mckeon.house.gov/about/purpose-mission-goals.shtml
- Costantini, Cristina. 2012. "U.S. Border Patrol Increases Use of Unmanned Drones for Surveillance." *Huffington Post*. May 1. Available online at: http://www.huffingtonpost.com/2012/05/01/us-border-patrolincrease n 1467196.html
- Cowen, Tyler. 2009. "Does Technology Drive the Growth of Government?" Mimeo, Available online: http://www.bcaplan.com/Cowentech.pdf
- Coyne, Christopher J. 2005. "The Institutional Prerequisites for Post-Conflict Reconstruction." *The Review of Austrian Economics*, 18(3/4): 325-342.
- Coyne, Christopher J. 2008. "The Politics of Bureaucracy and the Failure of Post-War Reconstruction." *Public Choice*, 135: 11-22.
- Coyne, Christopher J. and Abigail R. Hall. 2014. "Perfecting Tyranny: Foreign Intervention as Experimentation in State Control." *The Independent Review: A Journal of Political Economy* 19(2): 1-25.
- Coyne, Christopher J. and Adam Pellillo. 2011. "Economic Reconstruction Amidst Conflict: Insights from Afghanistan and Iraq." *Defence and Peace Economics*, 22(6): 627-643.
- Coyne, Christopher J. and Tyler Cowen. 2005. "Postwar Reconstruction: Some Insights from Public Choice and Institutional Economics." *Constitutional Political Economy*, 16: 31-48.

- Cratty, Carol. 2013. "FBI Uses Drones for Surveillance in U.S." CNN Politics. June 20. Available online at: http://www.cnn.com/2013/06/19/politics/fbidrones/index.html
- Cronin, Audrey Kurth. 2013. "Why Drones Fail: When Tactics Drive Strategy." *Foreign Affairs*, 29(4): 44-53.
- Crump, Catherine and Jay Stanley. 2013. "Why Americans are Saying No to Domestic Drones." *Slate*. February 11. Available online at: http://www.slate.com/articles/technology/future_tense/2013/02/domestic_surveill ance drone bans are sweeping the nation.html
- Customs and Border Protections Today. 2004. "Unmanned Aerial Vehicles Support Border Security." Available online: http://www.cbp.gov/xp/CustomsToday/2004/Aug/other/aerial vehicles.xml
- Dao, James. 2013. "Drone Pilots are Found to Get Stress Disorders Much as Those in Combat Do." *The New York Times*. February 23. Available online at: http://www.nytimes.com/2013/02/23/us/drone-pilots-found-to-get-stressdisorders-much-as-those-in-combat-do.html.
- Defense Industry Daily. 2013. "U.S. Will Sell Global Hawks—Will South Korea Buy?" *Defense Industry Daily*. January 2. Available online at: http://www.defenseindustrydaily.com/report-us-agrees-to-sell-global-hawks-tosouth-korea-05032/
- Department of Defense. 2004. "Elements of Defense Transformation." Available online at: www.dtic.mil/cgi-bin/GetTRDoc?AD=ADA450716
- Department of Homeland Security: Office of the Inspector General. 2012. "CBP's Use of Unmanned Aircraft Systems in the Nation's Border Security." Available online at: http://www.oig.dhs.gov/assets/Mgmt/2012/OIG_12-85_May12.pdf
- DeYoung, Karen and Scott Clement. 2013. "Many Americans Say the Afghan War isn't Worth Fighting." *The Washington Post*, July 26. Available online at: http://www.washingtonpost.com/world/national-security/many-americans-say-afghan-war-isnt-worth-fighting/2013/07/25/d0447d44-f559-11e2-aa2e-4088616498b4_story.html.
- Dipert, Randall R. 2012. "The Ethics of Cyber Warfare." *Journal of Military Ethics*, 9(4): 384-410.

- Dourado, Eli. 2013. "The Next Internet-like Platform for Innovation? Airspace (think drones)," *Wired*, April 23. Available online: http://www.wired.com/opinion/2013/04/then-internet-now-airspace-dont-stifle-innovation-on-the-next-great-platform
- Drake, Bruce. 2013. "Obama and Drone Strikes: Support but Questions at Home, Opposition Abroad." *Pew Research Center*. May 24. Available online at: http://www.pewresearch.org/fact-tank/2013/05/24/obama-and-drone-strikessupport-but-questions-at-home-opposition-abroad/.
- Drake, Bruce. 2014. "Plurality of Americans Support Current Level of Defense Spending." Pew Research Center. February 24. Available online at: http://www.pewresearch.org/fact-tank/2014/02/24/plurality-of-americans-supportcurrent-level-of-defense-spending/.
- Dreazen, Yochi J. 2011. "From Pakistan, With Love: The Technology Used to Monitor the Skies Over Waziristan is Coming to Your Hometown." *National Journal*, March 12. Available online at: http://www.nationaljournal.com/magazine/dronesmay-be-coming-to-your-hometown-20110313
- Drug Enforcement Administration. 2013. "DEA Programs: State and Local Task Forces." Available online at: http://www.justice.gov/dea/ops/taskforces.shtml
- Duncan, Thomas K. and Christopher J. Coyne. 2013a. "The Overlooked Costs of a Permanent War Economy: A Market Process Approach," *The Review of Austrian Economics*, forthcoming.
- Duncan, Thomas K. and Christopher J. Coyne. 2013b. "The Origins of the Permanent War Economy: A Consequence of Government, Not Capitalism," *Mimeo*.
- Dunne, J. Paul. 1995. "The Defense Industrial Base." pp. 399-430. In Keith Hartley and Todd Sandler (eds.). 1995. *Handbook of Defense Economics*. Amsterdam: Elsevier.
- *Economist.* 2012. "The Dronefather." *The Economist.* December 1. Available online at: http://www.economist.com/news/technology-quarterly/21567205-abe-karemcreated-robotic-plane-transformed-way-modern-warfare
- Ellickson, Robert C. 1987. "A Critique of Economic and Sociological Theories of Social Control," *Journal of Legal Studies* 16: 67-99.
- Entous, Adam and Julian E. Barnes. 2014. "U.S. Secretly Flying Drones Over Iraq." *The Wall Street Journal*. June 12. Available online at: http://online.wsj.com/articles/us-secretly-flying-unmanned-drones-over-iraq-1402605640.

- Entous, Adam. 2010. "Special Report: How the White House Learned to Love the Drone." *Reuters*. May 18. Available online at: http://www.reuters.com/article/2010/05/18/us-pakistan-drones-idUSTRE64H5SL20100518.
- Fetherston, Drew. 1998. "Raising Grumman: How Leroy Grumman and Jake Swirbul Build a High-Flying Company From the Ground Up." *Newsday*. April 19.
- Foust, Joshua. 2012. "Targeted Killing, Pro and Con: What to Make of U.S. Drone Strikes in Pakistan. *The Atlantic*. September 26. Available online at: http://www.theatlantic.com/international/archive/2012/09/targeted-killing-proand-con-what-to-make-of-us-drone-strikes-in-pakistan/262862/.
- Fox, J. Ronald, David G. Allen, Thomas C. Lassman, Walton S. Moody, and Philip L. Shiman. 2011. Defense Acquisition and Reform, 1960-2009: An Elusive Goal. Washington, D.C.: Center of Military History United States Army.
- Francis, David. 2013. "Death by Drones: Ate They Worth the Cost?" The Fiscal Times, February 7. Available online at: http://www.thefiscaltimes.com/Articles/2013/02/07/Death-by-Drones-Are-They-Worth-the-Cost.
- Friess, Jay. 2011. "Pentagon Report Highlights Flaws in the Fire Scout UAV." SoMD News. July 15. Available online at: http://www.somdnews.com/article/20110715/NEWS/707159855/1056/pentagonreport-highlights-flaws-in-fire-scout-uav&template=southernMaryland.
- Frontline. 2013. "Weapons: Drones (RPVs)." Available online at: http://www.pbs.org/wgbh/pages/frontline/gulf/weapons/drones.html
- Frontline. 2013. "Weapons: Drones (RPVs)." Available online at: http://www.pbs.org/wgbh/pages/frontline/gulf/weapons/drones.html
- Fulton, Sandra. 2014. "Police Hunger for Drones May be Growing, but So Are Privacy Concerns." American Civil Liberties Union Legislative Office, January 16. Available online at: https://www.aclu.org/blog/technology-and-liberty-nationalsecurity/police-hunger-drones-may-be-growing-so-are-privacy
- General Atomics. 2013. "Careers." Available online at: https://sjobs.brassring.com/TGWebHost/searchopenings.aspx?SID=^iu69OxQ8b DaUsP0ka7ltXnszQObA8tocit4O2vYQNoME21/Cyo9fZ/AVNEejOK/s.
- General Dynamics. 2013. "Locations." Available online at: http://www.gdit.com/About-Us/Locations/.
- Gertler, Jeremiah. 2012. "U.S. Unmanned Aerial Systems." Congressional Research Service Report. Available online at: http://www.fas.org/sgp/crs/natsec/R42136.pdf
- Gertler, Jeremiah. 2012. "U.S. Unmanned Aerial Systems." Congressional Research Service Report. Available online at: http://www.fas.org/sgp/crs/natsec/R42136.pdf.
- Gilmore, J. Michael. 2011. Director, Operational Test and Evaluation: RQ-4B Global Hawk Block 30 Operational Test and Evaluation Report. Operational Test and Evaluation. Available online at: http://pogoarchives.org/m/ns/pentagon-ot-and-eeval-rq-4b-global- hawk-20110526.pdf.
- Goebel, Greg. 2012. "Unmanned Aerial Vehicles: USA." Available online at: http://www.vectorsite.net/twuav.html
- Goodale, Gloria. 2013. "States Consider Drone Bans: Overreaction or Crucial for Privacy Rights?" *Christian Science Monitor*. February 6. Available online at: http://www.csmonitor.com/USA/Politics/2013/0206/States-consider-drone-bans-Overreaction- or-crucial-for-privacy-rights-video
- Gordon, Scott. 2002. Controlling the State: Constitutionalism from Ancient Athens to Today. Cambridge: Harvard University Press.
- Government Accountability Office. 2003. "Defense Budget: Tracking of Emergency Response Funds for the War on Terror." Available online at: http://www.gpo.gov/fdsys/pkg/GAOREPORTS-GAO-03-346/html/GAOREPORTS-GAO-03-346.htm.
- Guardian. 2012. "Drones by Country: Who Has All the UAVs?" *The Guardian*. August 3. Available online at: http://www.guardian.co.uk/news/datablog/2012/aug/03/drone-stocks-bycountry#data
- Hall, Abigail R. and Christopher J. Coyne. "The Political Economy of Drones." *Defence* and Peace Economics, 25(5): 445-460.
- Hall, Abigail R. and Christopher J. Coyne. 2013. "The Creeping Militarization of U.S. Domestic Policing." *The Independent Review*, 17(4): 485-504

- Harder, Donelle and Jake Heisten. "Inhofe, Manchin Announce Senate Unmanned Aerial Systems (UAS) Caucus." Senator James M. Inhofe Press Releases. September 27. Available online at: http://inhofe.senate.gov/public/index.cfm? FuseAction=PressRoom.PressReleases&ContentRecord_id=08e3e107-d52f-8498-6a8f-ba7c980631aa
- Harder, Donelle and Jake Heisten. 2012. "Inhofe, Manchin Announce Senate Unmanned Aerial Systems (UAS) Caucus." Senator James M. Inhofe Press Releases, September 27. Available online at: http://inhofe.senate.gov/public/index.cfm? FuseAction=PressRoom.PressReleases&ContentRecord_id=08e3e107-d52f-8498-6a8fba7c980631aa.
- Hartley, Keith and Todd Sandler (eds.). 1995. *Handbook of Defense Economics*. Amsterdam: Elsevier.
- Hartung, William. 2011. Prophets of War: Lockheed Martin and the Making of the Military-Industrial Complex. New York: Nation Books.
- Hennigan, W.J. 2013. "Global Hawk Drone Flied into Budget Battle between Pentagon, Congress." L.A. Times, December 6. Available online at: http://www.latimes.com/business/la-fi-spy-drone-lives-20131206-story.html.
- Higgs, Robert. 1987. Crisis and Leviathan: Critical Episodes in the Growth of American Government. New York: Oxford University Press.
- _____. 2004. *Against Leviathan: Government Power and a Free Society*. California: Independent Institute.
- . 2007. "Military-Economic Fascism: How Business Corrupts Government, and Vice Versa," *The Independent Review* 12(2): 299–316.
- _____. 2007. *Neither Liberty nor Safety: Fear, Ideology, and the Growth of Government.* California: Independent Institute.
- _____. 2012. Delusions of Power: New Explorations of State, War, and Economy. California: Independent Institute.
- Hirshleifer, Jack. 2001. The Dark Side of the Force: Economic Foundations of Conflict Theory. Cambridge, MA: Cambridge University Press.
- Hosenball, Mark. 2012. "Two-Thirds of Top al'Qaeda Leaders "Removed" Since 2009: Obama Aide. *Reuters*. December 18. Available online at: http://www.reuters.com/article/2012/12/18/us-usa-qaeda-officialidUSBRE8BH19O20121218.

- Hossein-Zadeh, Ismael. 2006. *The Political Economy of U.S. Militarism*. New York: Palgrave MacMillan.
- Hunsaker, J.C. 1954. *Elmer Ambrose Sperry*. Washington, D.C.: National Academy of Sciences.
- Ikeda, Sanford (1997). Dynamics of a Mixed Economy: Toward a theory of interventionism, London, UK: Routledge.

_____. 2005). "The Dynamics of Interventionism," *Advances in Austrian Economics*, 8: 21-57.

- International Human Rights and Conflict Resolution Clinic and Global Justice Institute. 2012, "Living Under Drones." Available online at: http://livingunderdrones.org/wp-content/uploads/2012/10/Stanford-NYU-LIVING-UNDER-DRONES.pdf
- International Human Rights and Conflict Resolution Clinic at Stanford Law School and Global Justice Clinic at NYU School of Law. 2012. "Living Under Drones: Death, Injury, and Trauma to Civilians from US Drone Practices in Pakistan." Available online at: <u>http://www.livingunderdrones.org/wp-</u> content/uploads/2013/10/Stanford-NYU-Living-Under-Drones.pdf.
- Isard, Walter. 1992. Understanding Conflict and the Science of Peace. Cambridge, MA: Blackwell.
- Israel Aircraft Industries. 2014. "Company Profile." Available online at: http://www.iai.co.il/2013/10285-en/CompanyInfo-CompanyProfile.aspx
- Jacoby, Henry. 1973. *The Bureaucratization of the World*. Los Angeles: University of California Press.
- Jeffrey, Terence P. 2012. "FAA Has Authorized 106 Government "Entities" to Fly Domestic Drones." *CNSNews*. July 20. Available online at: http://cnsnews.com/news/article/faa- has-authorized-106-government-entities-flydomestic-drones
- Jenks, Chris. 2010. "Law From Above: Unmanned Aerial Systems, Use of Force and the Law of Armed Conflict." *North Dakota Law Review*, 85: 650-670.
- John Warner National Defense Authorization Act for Fiscal Year 2007. Public Law 109-364, section 941.

- Johnston, Patrick B., Anoop K. Sarbahi. "The Impact of U.S. Drone Strikes on Terrorism in Pakistan and Afghanistan." Available online at: http://patrickjohnston.info/materials/drones.pdf.
- Jones, Seth G. 2014. "A Persistent Threat: The Evolution of al Qai'da and Other Salafi Jihadists." RAND Corporation. Available online at: http://www.rand.org/content/dam/rand/pubs/research_reports/RR600/RR637/RA ND_RR637.pdf.
- Jones, Seth G. and Martin C. Libicki. 2008. "How Terrorist Groups End: Lessons for Countering al Qa'ida. RAND Corporation. Available online at: https://www.rand.org/content/dam/rand/pubs/monographs/2008/RAND_MG741-1.pdf.
- Kilcullen, David and Andrew McDonald Exum. 2009. "Death From Above, Outrage Down Below." *New York Times*. May 16. Available online at: http://www.nytimes.com/2009/05/17/opinion/17exum.html?pagewanted=all&_r= 0
- Kilcullen, David and Andrew McDonald Exum. 2009. "Death From Above, Outrage Down Below." *New York Times*. May 16. Available online at: http://www.nytimes.com/2009/05/17/opinion/17exum.html?pagewanted=all&_r= 0.
- Kirzner, Israel M. (1985). *Discovery and the Capitalist Process*, Chicago, IL: University of Chicago Press.
- Koppl, Roger. and Leland B. Yeager. 1996. "Big Players and Herding in Asset Markets: The Case of the Russian Ruble." *Explorations in Economic History*, 33(3): 367–383.
- Kreps, Sarah and Micah Zenko. 2014. "The Next Drone Wars: Preparing for Proliferation." *Foreign Affairs*, March/April: 68-79.
- Kuhnhenn, Jim. 2008. "McCainL Cutting Casualties in Iraq is Primary Goal." *The Seattle Times*. June 11. Available online at: http://seattletimes.com/html/politics/2004470594_apmccainiraq.html
- Lewis, Michael W. 2013. "Drones: Actually the Most Humane Form of Warfare Ever: How Better Targeting and Surveillance Can Reduce the Number of Civilian Casualties." *The Atlantic*, August 21. Available online at: http://www.theatlantic.com/international/archive/2013/08/drones-actually-themost-humane-form-of-warfare-ever/278746/.

- Lockheed Martin. 2013. "Lockheed Martin Location Map." Available online at: http://www.lockheedmartin.com/content/dam/lockheed/data/corporate/documents/ Locati ons-Map-US.pdf.
- Lockheed Martin. 2013c. "100 Years of Accelerating Tomorrow." Available online at: http://www.lockheedmartin.com/us/100years.html
- Lubold, Gordon and John Reed. 2013. "This Drone Just Made History. But Now Where Does it Go?" *Foreign Policy*. July 10. Available online at: http://complex.foreignpolicy.com/posts/2013/07/10/this_drone_just_made_history but now where does it go.
- Lyon, David H. 2004. "A Military Perspective on Small Unmanned Aerial Vehicles." *IEEE Instrumentation and Measurement Magazine*, September: 27-31.
- Lyon, David H. 2004. "A Military Perspective on Small Unmanned Aerial Vehicles." *IEEE Instrumentation and Measurement Magazine*, September: 27-31.
- Madhani, Aamer. 2012. "Obama Announces Pentagon Budget Cuts." USA Today, January 5. Available online at: http://content.usatoday.com/communities/theoval/post/2012/01/obama-defensecuts/1#.UkrbqoasiSo.
- Madison, James. [1788] n.d. The Federalist No. 51. In *The Federalist*. New York: Modern Library.
- Mazzetti, Mark. 2013. The Way of the Knife. New York: The Penguin Press.
- McKeon, Buck. 2013. "Defense and National Security." Available online at: http://mckeon.house.gov/issues/issue/?IssueID=9831
- Melman, Seymour (ed). 1971. *The War Economy of the United States: Readings in the Military Industry and Economy*. New York: St. Martin's Press.
- Melman, Seymour. 1970. Pentagon Capitalism: The Political Economy of War. New York: McGraw-Hill.

_____. 1985. *The Permanent War Economy: American Capitalism in Decline*, New York: Simon & Schuster, Inc.

Merton, Robert K. 1940. "Bureaucratic Structure and Personality," *Social Forces* 18(4): 560-568.

Miller, Greg. 2012. "Drone Wars," Science 336(6083): 842-843.

- Miller, Greg. 2014. "Congress Balks at Drone Oversight Change." *Valley News*, January 16. Available online at: http://www.vnews.com/news/nation/world/10236792-95/congress-balks-at-drone-oversight-change.
- Millett, Allan R., Peter Maslowski, and William B. Feis. 2012. For the Common Defense: A Military History of the United States from 1607 to 2012. New York, NY: Free Press.
- Mises, Ludwig Von. [1929] (1996). *Critique of Interventionism,* Irvington-On-Hudson, NY: Foundation for Economic Education.

_____. [1949] (2007). *Human Action: A Treatise on Economics,* Indianapolis: IN: Liberty Fund,

Mobbs, Paul. 2014. "Drones, Cyberwarfare, and Democracy." *Action Atomic Weapons Eradication*. Available online at: http://actionawe.org/drones-cyberwarfare-anddemocracy/

Monmouth University Polling Institute. 2013. "National: U.S. Supports Unarmed Domestic Drones: But Public Prefers Requiring Court Orders First." Available online at: https://www.monmouth.edu/assets/0/32212254770/32212254991/32212254992/3 2212254994/32212254995/30064771087/409aecfb-3897-4360-8a05-03838ba69e46.pdf

Moorman, Robert W. 2009. "ISR: Manned vs. Unmanned." *Defense News*. June 2. Available online at: http://www.defensenews.com/article/20090602/C4ISR02/906020305/ISR-Manned-vs-unmanned.

- Murray, Patrick. 2012. "U.S. Supports Some Domestic Drone Use: But Public Registers Concerns About Own Privacy." Monmouth University Poll. Available online at: http://www.monmouth.edu/assets/0/32212254770/32212254991/32212254992/32 212254994/32212254995/30064771087/42e90ec6a27c40968b911ec51eca6000.p df.
- Musgrave, Shawn. 2014. "Every Drone Mission the FBI Admits to Flying." *Motherboard*. April 16. Available online at: http://motherboard.vice.com/read/every-drone-mission-the-fbi-admits-to-flying
- National Association of Engineers. 2014. "NAE Members." Available online at: http://www.nae.edu/MembersSection/Directory20412/31125.aspx?layoutChange= Print,LowGraphics

- National Aviation. 2013a. "Glenn Martin." National Aviation Hall of Fame. Available online at: http://www.nationalaviation.org/martin-glenn/
- National Defense Industrial Association. 2013. "History of the NDIA." Available online at: http://www.ndia.org/AboutUs/Pages/HistoryofNDIA.aspx
- National Museum of the United States Air Force. 2011. "Radioplane OQ-2A." Available online at: http://www.nationalmuseum.af.mil/factsheets/factsheet.asp?id=486
- NBC News. 2007. "Nov. 15 Democratic Debate Transcript." *NBC News*. November 16. Available online at: http://www.nbcnews.com/id/21836286/ns/politicsthe_debates/#.UTOCmDDqmSo
- New America Foundation. 2013. "Analysis: The Drone War in Pakistan." Available online at: http://natsec.newamerica.net/drones/pakistan/analysis
- Newcome, Laurence R. 2004. Unmanned Aviation: A Brief History of Unmanned Aerial Vehicles. Reston, VA: American Institute of Aeronautics and Astronautics, Inc.
- Newcome, Laurence R. 2004. Unmanned Aviation: A Brief History of Unmanned Aerial Vehicles. Reston, VA: American Institute of Aeronautics and Astronautics, Inc.
- Newcome, Laurence R. 2004. Unmanned Aviation: A Brief History of Unmanned Aerial Vehicles. Reston, VA: American Institute of Aeronautics and Astronautics, Inc.
- Niskanen, William A.1971. *Bureaucracy and Representative Government*. Chicago and New York City: Aldine, Atherton.
- North, Douglas C. 1990. *Institutions, Institutional Change, and Economic Performance*. New York: Cambridge University Press.
- Northrop Grumman. 2012a. "2012 Annual Report." Available online at: http://www.northropgrumman.com/AboutUs/AnnualReports/Documents/pdfs/201 2_noc_ar.pdf
- _____. 2012b. "Support Global Hawk." Available online at: http://web.archive.org/web/20120101031940/http://www.northropgrumman.com/s upportglobalhawk/index.html.
- . 2013. "Careers." Available online at: http://www.northropgrumman.com/Careers/DiscoverNorthropGrumman/Pages/Lo cations .aspx.

- O'Hanlon, Michael. 2011. "The National Security Industrial Base: A Crucial Asset of the United States, Whose Future May Be in Jeopardy." 21st Century Defense Initiative at Brookings. Available online at: http://www.brookings.edu/~/media/research/files/papers/2011/2/defense%200hanl on/02_defense_ohanlon.pdf
- Obama, Barack. 2013. "Speech on U.S. Drone and Counterterror Policy." May 23. Available online at: http://www.nytimes.com/2013/05/24/us/politics/transcript-ofobamas-speech-on-drone-policy.html?pagewanted=all&_r=0.
- Office of the Inspector General. 2012. "CBP's Use of Unmanned Aircraft Systems in the Nations' Border Security." *Department of Homeland Security*. Available online at: http://www.oig.dhs.gov/assets/Mgmt/2012/OIG_12-85_May12.pdf
- Olson, Mancur. 1965. *The Logic of Collective Action: Public Goods and the Theory of Groups*. Cambridge: Harvard University Press.
- Olvito, Jonathan. 2013. "Beyond the Fourth Amendment: Limiting Drone Surveillance through the Constitutional Right to Information Privacy." *Ohio State Law Journal*, 74(4): 669-702.
- Operation Border Star. 2014. "Operation Border Star Participants." Available online at: http://www.dps.texas.gov/PublicInformation/OpBorStarParticipants.pdf
- Paine, Thomas. 1775-6 [1894]. "Common Sense," In, *The Writings of Thomas Paine*, Moncure Daniel Conway (ed.), New York: The Knickerbocker Press, pp. 67-120.
- Palmer, Brian. 2012. "Is it Hard to Kill a Drone? Pakistan Threatens to Shoot The, Down, but Never Does." *Slate*. June 6. Available online at: http://www.slate.com/articles/news_and_politics/explainer/2012/06/cia_drone_pr ogram_is_it_hard_to_shoot_one_down_.html.
- Pappalardo, Joe. 2010. "The Future for UAVs in the U.S. Air Force." *Popular Mechanics*. February 26. Available online at: http://www.popularmechanics.com/technology/aviation/military/4347306.
- Parsch, Andreas. 2003. "Boeing BQ-7 *Aphrodite*." Directory of U.S. Military Rockets and Missiles. Available online at: http://www.designationsystems.net/dusrm/app1/bq-7.html
- . 2010. "Missile Index." Directory of U.S. Military Rockets and Missiles. Available online at: http://www.designationsystems.net/dusrm/app1/index.html#_Index_DronesAF

- Paul, Rand. 2013. "Sen. Rand Paul: My Filibuster was Just the Beginning." *The Washington Post*. March 8. Available online at: http://www.washingtonpost.com/opinions/sen-rand-paul-my-filibuster-was-just-the-beginning/2013/03/08/6352d8a8-881b-11e2-9d71-f0feafdd1394 story.html
- Pearson, Lee. 1969. "Developing the Flying Bomb." In Adrian O. Van Wyen (ed.) *Naval Aviation in WWI*. Washington, D.C.: Chief of Naval Operations. Available online at: http://www.history.navy.mil/branches/ww1av.htm
- Perry, Rick. 2008. "Texas Securing Our Nation's Border." Office of the Governor Rick Perry. Available online at: http://governor.state.tx.us/news/editorial/10224/
- Pew Research Center. 2013. "Balancing Civil Liberties and Security." December 3. Available online at: http://www.people-press.org/2013/12/03/public-sees-u-s-power-declining-as-support-for-global-engagement-slips/12-3-2013-34/.
- . 2014. "U.S. Foreign Policy: Key Data Points from Pew Research." Pew Research Center. January 6. Available online at: http://www.pewresearch.org/key-datapoints/u-s-foreign-policy-key-data-points/.
- Poast, Paul. 2006. The Economics of War. New York: McGraw-Hill Irwin.
- Porter, Bruce D. 1994. War and the Rise of the State: The Military Foundations of Modern Politics. New York: The Free Press.
- PR Newswire. 2012. "Teal Group Predicts Worldwide UAV Market Will Total \$89 Billion in its 2012 UAV Market Profile and Forecast." *PR Newswire*. April 11. Available online at: http://www.prnewswire.com/news-releases/teal-grouppredicts-worldwide-uav-market-will-total-89-billion-in-its-2012-uav-marketprofile-and-forecast-147008115.html
- Pringle, Fiona. 2012. "Navy's Fire Scout UAV Compromises BAE's Precision Weapons Upgrade." *Defense Report*. October 12. Available online at: http://defencereport.com/navys-fire-scout-uav-compromises-baes-precisionweapons-upgrade/.
- Program for Public Consultation. 2012. "Consulting the American People on National Defense Spending." Available online at: https://s3.amazonaws.com/s3.documentcloud.org/documents/355447/defensebudget-survey-full-results.pdf.
- Public Broadcasting Station. 2013a. "William Boeing." Available online: http://www.pbs.org/kcet/chasingthesun/innovators/wboeing.html

Pursell, C. 1972. The Military Industrial Complex. New York: Row Publishers.

- Ratnesar, Romesh. 2013. "Five Reasons Why Drones are Here to Stay." *Bloomberg Businessweek*. May 23. Available online at: http://www.businessweek.com/articles/2013-05-23/five-reasons-why-drones-arehere-to-stay.
- Reed, John. "Predator Drones 'Useless' in Most Wars, Top Air Force General Says." *Foreign Policy*. September 19. Available online at: http://complex.foreignpolicy.com/posts/2013/09/19/predator_drones_useless_in_ most_wars_top_air_force_general_says.
- Reuter, Peter, Gordon Crawford, and Jonathan Cave. "Sealing the Borders: The Effects of Increased Military Participation in Drug Interdiction." RAND National Defense Research Institute. Available online at: http://www.rand.org/content/dam/rand/pubs/reports/2007/R3594.pdf
- Rockwell, Mark. 2012. "States Anxious on FAA's UAV Test Sites." *GSN News*. September 25. Available online at: http://www.gsnmagazine.com/node/27370?c=airport_aviation_security
- Rogers, Simon. 2012. "US Drone Strikes Listed and Detailed in Pakistan, Somalia and Yemen." *The Guardian*. August 2. Available online at: http://www.theguardian.com/news/datablog/2012/aug/02/us-drone-strikes-data.
- Roggio, Bill and Alexander Mayer. 2013. "Charting the Data for US Airstrikes in Pakistan, 2004-2013." *The Long War Journal*. Available online: http://www.longwarjournal.org/pakistan-strikes.php
- Roston, Aram. 2013. "Will Congress Let the Air Force Abandon the Global Hawk?" *Defense News*. January 24. Available online at: http://www.defensenews.com/article/20130624/C4ISR01/307010011/Will-Congress-Let- Air-Force-Abandon-Global-Hawk-.
- Rothbard, Murray N. 1970. *Power and Market*. Menlo Park, CA: Institute for Humane Studies.
- Saletan, William. 2013. "In Defense of Drones: They're the Worst Form of War, Except for All the Others." *Slate*. February 19. Available online at: http://www.slate.com/articles/health_and_science/human_nature/2013/02/drones_ war_and_civilian_casualties_how_unmanned_aircraft_reduce_collateral.html.

- Sandler, Todd and Keith Hartley. 1995. *The Economics of Defense*. New York: Cambridge University Press.
- Scott, Lloyd N. 1920. *Naval Consulting Board of the United States*. Washington, D.C.: Government Printing Office.
- Sengupta, Somini. 2013. "Predator Drones Keep an Eye on the Border, Documents Show." New York Times. February 28. Available online at: http://bits.blogs.nytimes.com/2013/02/28/predator-drones-keep-an-eye-on-theborder-documents-show/
- Sengupta, Somini. 2013. "U.S. Border Agency Allows Others to Use Its Drones." *New York Times*. July 3. Available online at: http://www.nytimes.com/2013/07/04/business/us-border-agency-is-a-frequentlender-of-its-drones.html?pagewanted=all&_r=0
- Seybolt, Taylor B. 2008. *Humanitarian Military Intervention: The Conditions for Success and Failure*. New York: Oxford University Press.
- Seybolt, Taylor B. 2008. *Humanitarian Military Intervention: The Conditions for Success and Failure*. New York: Oxford University Press.
- Sharkey, Noel. 2010. "Saying 'No!' to Lethal Autonomous Targeting." Journal of Military Ethics, 9(4): 369-383.
- Shelsby, Ted. 1991. *The Baltimore Sun*, March 2. Available online at: http://articles.baltimoresun.com/1991-03-02/business/1991061100_1_rpv-aaidrones
- Sherfinski, David. 2013. "Rumsfeld: 'There is Clearly Going to be a Use for Drones."" *The Washington Times*. May 23. Available online at: http://www.washingtontimes.com/news/2013/may/23/rumsfeld-there-clearlygoing-be-use-drones/.
- Sia, H.P. and Alexander Cohen. 2013. "The Drone that Wouldn't Die: How a Defense Contractor Bested the Pentagon." *The Atlantic*. July 16. Available online at: http://www.theatlantic.com/politics/archive/2013/07/the-drone-that-wouldnt-diehow-a- defense-contractor-bested-the-pentagon/277807/.
- Sifton, John. 2012. "A Brief History of Drones." *The Nation*, February 7. Available online at: http://www.thenation.com/article/166124/brief-history-drones#.

- Sifton, John. 2012. "A Brief History of Drones." *The Nation*. February 7. Available online at: http://www.thenation.com/article/166124/brief-history-drones#
- Simon, Herbert A. 1997. *Administrative Behavior*, 4th ed. New York, NY: The Free Press.
- Singer, Peter W. 2009. *Wired for War: The Robotics Revolution and Conflict in the 21st Century*. New York: Penguin Publishers.
- Smith, Adam, Richard Wagner, and Bruce Yandle. 2011. "A Theory of Entangled Political Economy, with Application to TARP and NRA." *Public Choice* 148: 45-66.
- Smith, Megan and James Igoe Walsh. 2013. "Do Drone Strikes Degrade Al Qaeda? Evidence From Propaganda Output." *Terrorism and Political Violence*, 25: 311-327.
- Smith, Ron P. 2009. *Military Economics: The Interaction of Power and Money*. New York: Palgrave Macmillan.
- Smithsonian National Museum. 2013a. "Katydid Drone." Available online at: http://airandspace.si.edu/collections/artifact.cfm?id=A19660162000
- _____. 2013b. "Northrop P-61C Black Widow." Available online at: http://airandspace.si.edu/collections/artifact.cfm?id=A19510044000
- Springer, Paul. 2013. *Military Robots and Drones*. Santa Barbara, California: ABC-CLIO Publishing.
- Stokes, Bruce. 2014. "Americans' Foreign Policy Priorities for 2014. PEW Research Center. Available online at: http://www.pewresearch.org/facttank/2013/12/31/americans-foreign-policy-priorities-for-2014/.
- Strategy and Policy. Bloomington, IN: Indiana University Press.
- Strawser, Bradley J. 2010. "Moral Predators: The Duty to Employ Uninhabited Aerial Vehicles." *Journal of Military Ethics* 9(4): 342-368.
- Strawser, Bradley J. 2010. "Moral Predators: The Duty to Employ Uninhabited Aerial Vehicles." *Journal of Military Ethics*, 9(4): 342-368.
- Swaine, Jon. 2013. "Barack Obama 'Has Authority to Use Drone Strikes to Kill Americans on US Soil." *The Telegraph*. March 6. Available online at:

http://www.telegraph.co.uk/news/worldnews/barackobama/9913615/Barack-Obama-has-authority-to-use-drone-strikes-to-kill-Americans-on-US-soil.html

- Swift, Christopher. 2012. "The Drone Blowback Fallacy." Foreign Affairs. July 1. Available online at: http://www.foreignaffairs.com/articles/137760/christopherswift/the-drone-blowback-fallacy.
- Teal Group. 2012. "Teal Group Predicts Worldwide UAV Market Will Total \$89 Billion in its 2013 UAV Market Profile and Forecast." Available online at: http://tealgroup.com/index.php/about-teal-group-corporation/press-releases/94-2013-uav-press-release.
- Thompson, Loren. 2014. "U-2 vs. Global Hawk: Why Drones Aren't Always the Answer to Every Military Need." *Forbes*. February 20. Available online at: http://www.forbes.com/sites/lorenthompson/2014/02/20/u-2-vs-global-hawk-whydrones-arent-the-answer-to-every-military-need/.
- Thompson, Richard M. 2013. "Drones in Domestic Surveillance Operations: Fourth Amendment Implications and Legislative Responses." Congressional Research Service. Available online at: http://www.fas.org/sgp/crs/natsec/R42701.pdf

Thruelson, Richard. 1976. The Grumman Story. New York: Praeger Publishers.

- Tomiuc, Eugen. 2012. "Drones—Who Makes Them and Who Has Them?" *Radio Free Europe Radio Liberty*. January 31. Available online at: http://www.rferl.org/content/drones_who_makes_them_and_who_has_them/2446 9168.html
- Tomiuc, Eugen. 2012. "Drones—Who Makes Them and Who Has Them?" *Radio Free Europe Radio Liberty*. January 31. Available online at: http://www.rferl.org/content/drones_who_makes_them_and_who_has_them/2446 9168.html.
- Toscano, Michael. 2011. "Unmanned Aircraft Systems: An Industry Perspective." Grand Forks, North Dakota. Available online at: http://www.slideshare.net/lhfang/drone-lobby-presentation.
- Treadwell, Terry. 1990. *Ironworks: Grumman's Fighting Aeroplanes*. Shrewsbury, UK: Afterlife Publishers.

Tullock, Gordon. 2005 [1965]. Bureaucracy. Indianapolis: Liberty Fund.

. 2008. "Public Choice." In Steven N Durlauf and Lawrence E. Blume (eds.), *The New Pelgrave Dictionary of Economics*, second edition. New York: Pelgrave Macmillan.

- United States Department of Defense. 2005. "Unmanned Aircraft Systems Roadmap 2005-2030." Available online at: https://www.fas.org/irp/program/collect/uav_roadmap2005.pdf
- . 2009. "2009-2034 Unmanned Systems Integrated Roadmap." Available online at: http://www.dtic.mil/cgibin/GetTRDoc?Location=U2&doc=GetTRDoc.pdf&AD=ADA522247
- . 2013. "Unmanned Systems Integrated Roadmap FY 2013-2038. Available online at: http://www.defense.gov/pubs/DOD-USRM-2013.pdf.
- United States National Defense Authorization Act for 2007. Available online at: http://www.gpo.gov/fdsys/pkg/PLAW-109publ364/pdf/PLAW-109publ364.pdf.
- United States National Defense Authorization Act for 2013. Available online at: http://www.gpo.gov/fdsys/pkg/BILLS-112hr4310enr/pdf/BILLS-112hr4310enr.pdf.
- United States Navy. 2009. "RQ-2A Pioneer Unmanned Aerial Vehicle." Available online at: http://www.navy.mil/navydata/fact_display.asp?cid=1100&tid=2100&ct=1
- Unmanned Aerial Vehicle Systems Association. 2013. "Advantages of UAS." Available online at: http://www.uavs.org/advantages
- USA Spending. 2013. Various Pages. Available online at: http://www.usaspending.gov/.
- Wagner, Richard. 2009a. Fiscal Sociology and the Theory of Public Finance: An Exploration Essay. Northampton, MA: Edward Elgar Publishing.
- . 2009b. "Property, State, and Entangled Political Economy." In Wolf Schafer, Andrea Schneider, and Tobias Thomas (eds.), Markets and Politics: Insights from a Political Economy Perspective. Marburg: Metropolis.
- Waldman, Paul. 2013. "Game of Drones." *The American Prospect*. February 14. Available online at: http://prospect.org/article/game-drones
- Watts, Barry D. 2008. "The U.S. Defense Industrial Base: Past, Present, and Future." Center for Strategic and Budgetary Assessments. Available online at: http://www.csbaonline.org/wp-content/uploads/2011/02/2008.10.15-Defense-Industrial-Base.pdf

Weigley, Russell F. 1960. The American Way of War: A History of United States Military

- Weingast, Barry. 1995. "The Economic Role of Political Institutions: Market-Preserving Federalism and Economic Development." Journal of Law, Economics, and Organization 11(1): 1-31.
- Williams, Brian Glyn. 2013. *Predators: The CIA's Drone War on al Qaeda*. Dulles, Virginia: Potomac Books, Inc.
- Wolverton, Joe. 2012. "Will Police Drones Destroy the Fourth Amendment?" *The New American*. November 6. Available online at: http://www.thenewamerican.com/usnews/constitution/item/13542-will-policedrones-destroy-the-fourth-amendment
- Wolverton, Joe. 2013. "Obama Orders Drone Strikes, Killing of "Suspected Militants" in Yemen. *The New American*. May 21. Available online at: http://www.thenewamerican.com/world-news/asia/item/15453-obama-ordersdrone-strikes-killing-of-6-suspected-militants-in-yemen.
- _____. 2014. "First Man Arrested by Aid of Drone Convicted in North Dakota." *The New American*. February 1. Available online at: http://www.thenewamerican.com/usnews/constitution/item/17534-first-man-arrested-by-aid-of-drone-convicted-in-north-dakota
- Zenko, Micah. 2012. "10 Things You Didn't Know About Drones." *Foreign Policy*. February 27. Available online at: http://www.foreignpolicy.com/articles/2012/02/27/10_things_you_didnt_know_a bout_drones.
- Zucchino, David. 2010. "War Zone Drone Crashes Add Up." *Los Angeles Times*. July 6. Available online at: http://articles.latimes.com/2010/jul/06/world/la-fg-dronecrashes-20100706.
- Zuckerman, Josh. 2013. "Why Do We Use Drones?" *Penn Political Review*. July 18. Available online at: http://pennpoliticalreview.org/archives/5027

BIOGRAPHY

Abigail R. Hall is the JIN Ph.D. Fellow in economics at the Mercatus Center at George Mason University, a Research Fellow with the Independent Institute, an Institute for Humane Studies Fellow, and PhD candidate at George Mason University in Fairfax, Virginia. She graduated with a B.A. in economics and business administration with an additional concentration in mathematics from Bellarmine University in Louisville, Kentucky.

Her broader research interests include Austrian Economics, Political Economy, Defense and Peace Economics, and Institutions and Economic Development. Her work includes topics surrounding the U.S. military and national defense, including, domestic police militarization, arm sales, weapons as foreign aid, the cost of military mobilization, and the political economy of military technology.

She enjoys running, swimming and being outside in her spare time. She likes football and loves going to hockey games with her fiancée. She cooks and can bake like nobody's business! Before starting graduate school she worked as a dance instructor teaching tap, jazz, and ballet and has 18 years experience in all three genres.